UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

Current Report
Pursuant to Section 13 or 15(d) of the
Securities Exchange Act of 1934

May 7, 2021

Date of Report (Date of earliest event reported)

INDUS REALTY TRUST, INC.

(Exact name of registrant as specified in charter)

| <u>Maryland</u> | | <u>06-0868496</u> | |
|---|--------------------------------|--|--|
| (State or other jurisdiction of incorporati | ion) | (IRS Employer Identification No.) | |
| (Commission File Number) | | <u>1-12879</u> | |
| 641 Lexington Avenue, New York, Ne (Address of principal executive offices) | w York | 10022 (Zip Code) | |
| Registrant's Telephone Number, includi | ng Area Code | <u>(212) 218-7910</u> | |
| (Former name | e or former address, if change | ed since last report) | |
| Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions: | | | |
| ☐ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425) | | | |
| ☐ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12) | | | |
| □ Pre-commencement communications pursuant to Rule 14d-2 (b) under the Exchange Act (17 CFR 240.14d-2(b)) | | | |
| ☐ Pre-commencement communications pursuant to Rule 13e-4 (c) under the Exchange Act (17 CFR 240.13e-4(c)) | | | |
| Securities registered or to be registered pursuant to Section 12(b) of the Act: | | | |
| Title of each class Common Stock, \$0.01 par value per share | Trading Symbol(s) INDT | Name of each exchange on which registered The Nasdaq Stock Market LLC | |
| Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Ac of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter). | | | |
| | | Emerging growth company □ | |
| If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box | | | |

Item 1.01 Entry into a Material Definitive Agreement.

On May 7, 2021, Old Statesville Road, LLC ("Old Statesville"), a wholly owned subsidiary of INDUS Realty Trust, Inc. ("INDUS" or the "Company"), entered into a construction loan agreement (the "2021 JPM Construction Loan") with JPMorgan Chase Bank, N.A. ("JPM") to provide a portion of the funds for the Company's development of a build-to-suit last-mile industrial/logistics facility of approximately 141,000 square feet in Charlotte, North Carolina (the "Charlotte Build-to-Suit") for Amazon, which entered into a 15 year lease with INDUS for the Charlotte Build-to-Suit. Total borrowings under the JPM Construction Loan will be the lesser of \$28.4 million or 67.5% of the project cost (as defined in the 2021 JPM Construction Loan) of the Charlotte Build-to-Suit. The term of the 2021 JPM Construction Loan is May 7, 2023, with a one-year extension at the Company's option. The interest rate under the 2021 JPM Construction Loan, to be adjusted monthly, is one-month LIBOR plus 1.65%, reduced to one-month LIBOR plus 1.40% upon completion of the Charlotte Build-to-Suit and commencement of rental payments by Amazon.

Upon execution of the 2021 JPM Construction Loan, Old Statesville was required to pay a fee in the amount of \$127,800.

The 2021 JPM Construction Loan contains customary representations and warranties, affirmative and negative covenants, and events of default that entitle JPM to cause the borrowings under the 2021 JPM Construction Loan to become immediately due and payable.

The foregoing description of the 2021 JPM Construction Loan is subject to and qualified in its entirety by reference to the full text of the 2021 JPM Construction Loan, a copy of which is filed herewith as Exhibit 10.1 and is incorporated herein by reference.

Item 2.02 Results of Operations and Financial Condition.

On May 10, 2021, INDUS issued a press release announcing its results of operations for the month of December 2020 and the quarter ended March 31, 2021.

Item 2.03. Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement

The information set forth in Item 1.01 above is incorporated by reference into this Item 2.03.

Item 7.01 Regulation FD Disclosure.

On May 10, 2021, INDUS issued a press release announcing: (a) its results of operations for the month of December 2020 and its fiscal 2021 first quarter; (b) the completion of the 2021 JPM Construction Loan; and (c) that on May 7, 2021, its Board of Directors declared a quarterly dividend on the Company's common stock of \$0.15 per share payable on June 30, 2021 to holders of record as of the close of business on June 16, 2021.

A copy of the Company's May 10, 2021 press release is attached hereto as Exhibit 99.1. The information contained and referenced in this Item 7.01, including the press release attached as Exhibit 99.1, is being "furnished" and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of

Section 18 of the Exchange Act. The information in this Item 7.01 shall not be incorporated by reference into any registration statement or other document pursuant to the Securities Act of 1933, as amended (the "Securities Act"), or into any filing or other document pursuant to the Exchange Act, except as otherwise expressly stated in any such filing.

Item 9.01. Financial Statements and Exhibits

- (d) Exhibits
- 10.1 <u>Construction Loan Agreement dated as of May 7, 2021 by and among Riverbend Old</u> Statesville, LLC as Borrower, and JPMorgan Chase Bank, N.A., as Lender
- 99.1 The Company's Press Release dated May 10, 2021 (attached hereto)
- 104 Cover Page Interactive Data File (formatted as inline XBRL and contained in Exhibit 101)

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

INDUS REALTY TRUST, INC.

By: /s/ Anthony J. Galici
Anthony J. Galici
Executive Vice President and Chief
Financial Officer

Date: May 10, 2021

Exhibit 10.1

REAL ESTATE BANKING

Construction Loan Agreement

DATED AS OF May 7, 2021

BY AND AMONG

RIVERBEND OLD STATESVILLE, LLC, a North Carolina limited liability company,

AS BORROWER,

AND

JPMORGAN CHASE BANK, N.A.,

AS LENDER

J.P.Morgan

TABLE OF CONTENTS

| ARTICLE I: DEFINED TERMS | 1 |
|--|----|
| ARTICLE II: LOAN TERMS | 12 |
| ARTICLE III: GENERAL COVENANTS; CONSTRUCTION | 18 |
| ARTICLE IV: REPRESENTATIONS AND WARRANTIES | 25 |
| ARTICLE V: INSURANCE | 27 |
| ARTICLE VI: DEFAULTS AND REMEDIES | 30 |
| ARTICLE VII: MISCELLANEOUS | 34 |

EXHIBITS

Exhibit A -- Legal Description

Exhibit B -- Closing Conditions

Exhibit C -- Advances

Exhibit D -- Litigation against Borrower, the Property or Guarantor

Exhibit E -- Borrower's Compliance Certificate

Exhibit F -- Guarantor's Compliance Certificate

Exhibit G -- Budget

J.P.Morgan

CONSTRUCTION LOAN AGREEMENT

This Construction Loan Agreement (the "<u>Agreement</u>") dated as of this 7th day of May, 2021, by and between Riverbend Old Statesville, LLC, a North Carolina limited liability company ("<u>Borrower</u>"), and JPMorgan Chase Bank, N.A., a national banking association ("Lender").

RECITALS

- A. Borrower owns all of that certain real property located in the City of Charlotte, County of Mecklenburg and State of North Carolina more particularly described in <u>Exhibit A</u> (the "<u>Land</u>"), and Borrower proposes to construct or cause to be constructed upon the Land certain improvements consisting of 141,360 square feet in accordance with the Plans and Specifications and as required under the Amazon Lease (as defined below); and
- B. Borrower has requested, and Lender has agreed to provide, financing to Borrower for the construction of such improvements on the terms and conditions herein.

Now, therefore, the parties agree as follows:

ARTICLE I: DEFINED TERMS

1.01 **Definitions.** The following terms shall have the following meanings:

"15% Equity Contribution Requirement" means the contribution of capital by Borrower into the Project of one or more of the following: cash (excluding land costs), unencumbered readily marketable assets, approved out-of-pocket development expenses ("Cash Equity") and the as-is appraised value of any contributed real property and improvements as determined by Lender based on its review of the Appraisal (accepted pursuant to Exhibit C), in each case as approved by Lender, that is equal to at least fifteen percent (15%) of the As-Completed Appraised Value of the Project as of the Closing Date. For the avoidance of doubt, the 15% Equity Contribution Requirement is not a separate equity requirement, but is a part of Borrower's total equity requirements herein.

"Advance" means a disbursement by Lender of any principal of the Loan.

"Affiliate" means, with respect to a specified Person, another Person that directly, or indirectly, Controls or is Controlled by or is under common Control with the Person specified.

"Amazon Guaranty" means that certain Limited Parent Guaranty dated effective as of March 23, 2021, executed by Amazon Parent in favor of Borrower, as the same may be amended, restated, amended and restated, supplemented or otherwise modified from time to time in accordance with the terms hereof.

"Amazon Lease" means that certain Lease Agreement dated effective as of March 23, 2021 by and between Borrower, as landlord, and Amazon, as tenant, as the same may be amended, restated, amended and restated, supplemented or otherwise modified from time to time in accordance with the terms hereof.

"Amazon Parent" means Amazon.com Inc., a Delaware corporation.

"Amazon Tenant" means Amazon.com Services LLC, a Delaware limited liability company.

"Annual Debt Service" means, as of any date of determination, annual debt service on a loan with a principal amount equal to the Commitment on such date of determination, assuming (a) a fixed rate of interest per annum equal to the greatest of (i) the Interest Rate as of such date of determination, (ii) the annual yield to maturity as

of such date of determination of a ten (10) year United States Treasury Note in the amount of such loan plus two and one-quarter of one percent (2.25%) per annum, or (iii) six percent (6.0%) per annum, and (b) mortgage-style amortization of such loan over a period of thirty (30) years.

"Anti-Corruption Laws" means all laws, rules, and regulations of any jurisdiction applicable to Borrower or any of its Affiliates from time to time concerning or relating to bribery or corruption.

"Applicable Margin" means 0.65% with respect to the CB Floating Rate, 1.65% with respect to the LIBOR Rate, and 1.65% with respect to the Daily Floating LIBOR Rate; provided, however, that the Applicable Margin shall be reduced to 0.40% with respect to the CB Floating Rate, 1.40% with respect to the LIBOR Rate, and 1.40% with respect to the Daily Floating LIBOR Rate, in each case, upon satisfaction of the following conditions, as determined by Lender in its reasonable discretion: (i) Substantial Completion of the Improvements has occurred; (ii) Amazon has accepted the Project in accordance with the terms and conditions of the Amazon Lease, (iii) the Amazon Lease has commenced and no default or event of default has occurred and is continuing thereunder, and (iv) no Default or Unmatured Default has occurred and is continuing.

"Appraisal" means a written appraisal of the market value of the Property that (i) has been independently prepared by a qualified appraiser directly engaged by Lender, (ii) complies with all applicable federal and state laws and regulations dealing with appraisals or valuations of real property, and (iii) has been reviewed as to form and content and approved by Lender, in its reasonable discretion.

"Appraised Value" means the "as-stabilized" value of the Property as determined by Lender based upon its review of the Appraisal.

"Approved Fund" means any Person (other than a natural person) that is engaged in making, purchasing, holding or investing in bank loans and similar extensions of credit in the ordinary course of its business and that is administered or managed by (a) Lender, (b) an Affiliate of Lender or (c) an entity or an Affiliate of an entity that administers or manages Lender.

"Approved Lease" means (a) the Amazon Lease, and (b) any other Lease specifically approved by Lender in writing.

"Architect" means Prime Engineering, Inc., a Georgia professional corporation.

"Architect's Certificate" means a certificate of Architect in form and substance acceptable to Lender.

"Architect's Contract" means the Agreement for Consulting Services between GLNCM, LLC and Architect dated as of November 16 2020, relating to the design of the Improvements, as amended from time to time.

"Architect Subordination Agreement" means the Consent to Assignment of Agreement with Architect and of Plans and Specifications dated as of May 5, 2021, executed by Architect in favor of Lender, as amended from time to time.

"<u>As-Completed Appraised Value</u>" means the "as completed" value, as determined by Lender based upon its review of the Appraisal of the Property accepted pursuant to <u>Exhibit C</u>.

"Balancing Deposit" has the meaning in Section 2.02(e).

J.P.Morgan

"BAQ" means that certain Borrower Administrative Questionnaire Form executed by Borrower in favor of Lender for this Loan.

"Beneficial Ownership Certification" means a certification regarding beneficial ownership or control as required by the Beneficial Ownership Regulation.

"Beneficial Ownership Regulation" means 31 C.F.R. Section 1010.230.

"Borrower's Equity Requirement" means an amount equal to all equity contributed to the Project described in Exhibit C.

"Borrowing Date" means a date on which an Advance is made under this Agreement.

"Borrower Financing Statement" means the UCC financing statement covering the security interests in personal property granted by Borrower to Lender in the Loan Documents for filing with the Secretary of State of the State of North Carolina.

"Borrower's Funds Account" has the meaning in Section 2.02(e).

"Borrowing Request" means a request by Borrower for an Advance in accordance with Section 2.02(c) and Exhibit C.

"Budget" means the line item breakdown of all Project Costs incurred or estimated to be incurred with respect to the Improvements attached as Exhibit G.

"Business Day" means any day that is not a Saturday, Sunday or other day on which commercial banks in New York City are authorized or required by law to remain closed; provided that, when used in connection with the LIBOR Rate, the term "Business Day" will also exclude any day on which banks are not open for general business in London.

"CB Floating Rate" means, for any day, the Prime Rate on such day; provided that the CB Floating Rate for any day shall never be less than the LIBOR Rate plus 2.50% on such day (or if such day is not a Business Day, the immediately preceding Business Day). For purposes of this definition the LIBOR Rate shall never be less than zero. Any change in the CB Floating Rate shall be effective from and including the effective date of such change.

"Closing Date" is the earlier of the date of the Initial Advance, or the date all the conditions to the closing of the Loan have been satisfied, including recordation, or Lender's authorization of recordation, of the Security Instrument.

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

"Commitment" means, the commitment of Lender to make the Loan in an amount not exceeding \$28,400,000.00, as such amount may be reduced from time to time pursuant to the terms of this Agreement.

"Commodity Exchange Act" means the Commodity Exchange Act (7 U.S.C. § 1 et seq.), as amended from time to time, and any successor statute.

"Completion Date" means the earlier of January 31, 2022, as may be extended as a result of the occurrence of Force Majeure Causes.

"Completion Guaranty" means the Completion Guaranty dated the same date as this Agreement executed by Guarantor in favor of Lender, as amended from time to time.

"Construction Contract" means that certain Construction Manager Contractor Agreement dated March 17, 2021, by and between Borrower, GLNCM, LLC, as Construction Manager, and the Contractor for construction of the Improvements, as amended from time to time.

"Construction Schedule" means a construction schedule for the Improvements in form and substance satisfactory to Lender.

"Contractor" means Intercom Building Company, LLC, a North Carolina limited liability company.

"Contractor Subordination Agreement" means the Consent to Assignment of Construction Agreement dated as of April 16, 2021, executed by Contractor in favor of Lender, as amended from time to time.

"Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlling" and "Controlled" have meanings correlative thereto.

"<u>Daily Floating LIBOR Rate</u>" means for any day the rate for borrowed funds in the interbank market in London in United States dollars for maturities of one month, rounded upwards if necessary to the next 1/16 of 1%, as determined at approximately 11:00 a.m. London time on that day. The Daily Floating LIBOR Rate for any day that is not a Business Day shall be the Daily Floating LIBOR Rate determined on the immediately preceding Business Day. Such rate shall be determined by Lender based upon the London Interbank Offered Rate as administered by the ICE Benchmark Administration (or any other entity that takes over administration of such rate for United States dollars). Any Daily Floating LIBOR Rate less than 0.0% shall be deemed to be 0.0% for purposes of this Agreement.

"<u>Debt Service Coverage Ratio</u>" means, as of any determination date, the ratio of NOI as of such determination date to Annual Debt Service as of such determination date.

"Debt Yield" means the ratio of (i) NOI to (ii) the then-outstanding Commitment.

"<u>Debtor Relief Laws</u>" means any applicable liquidation, conservatorship, bankruptcy, moratorium, rearrangement, insolvency, fraudulent conveyance, reorganization, or similar laws affecting the rights, remedies or recourse of creditors generally, including the United States Bankruptcy Code and all amendments, as are in effect from time to time during the term of the Loan.

"Default" has the meaning in Section 6.01.

"Default Rate" has the meaning in Section 2.04(b).

"<u>Developer Fee</u>" means the fee payable to Borrower, as developer, under the Development Agreement and identified in the Budget.

"<u>Development Agreement</u>" means that certain Development Agreement with an effective date of March 23, 2021 by and between Borrower and Amazon, as amended from time to time.

"Draw Package" has the meaning in Exhibit C.

J.P.Morgan

"<u>Electronic Signature</u>" means an electronic sound, symbol, or process attached to, or associated with, a contract or other record and adopted by a Person with the intent to sign, authenticate or accept such contract or record.

"Employee Benefit Plan" means an employee benefit plan as defined in Section 3(3) of ERISA, maintained, sponsored by or contributed to by Borrower or any ERISA Affiliate.

"Engineer" means Oak Engineering, PLLC, a North Carolina professional limited liability company.

"Engineer's Contract" means that certain Agreement for Consulting Services between GLNCM, LLC and Engineer dated as of November 13, 2020, as amended from time to time.

"Engineer Subordination Agreement" means the Consent To Assignment Of Agreement With Engineer And Of Plans And Specifications dated as of April 16, 2021, executed by Engineer in favor of Lender, as amended from time to time.

"Environmental Indemnity Agreement" means that certain Environmental Indemnity Agreement dated the same date as this Agreement executed by Borrower and Guarantor in favor of Lender, as amended from time to time.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time and the rules and regulations promulgated thereunder.

"ERISA Affiliate" means any trade or business (whether or not incorporated) that, together with Borrower, is treated as a single employer under Section 414(b) or (c) of the Code or Section 4001(14) of ERISA or, solely for purposes of Section 302 of ERISA and Section 412 of the Code, is treated as a single employer under Section 414 of the Code.

"Extension" means each extension of the Maturity Date pursuant to Section 2.07.

"Flood Insurance Requirements" means Lender has received evidence indicating whether the Improvements or any part thereof are or will be located within a one hundred year flood plain or other area identified by Lender as having high or moderate risk of flooding or identified as a special flood hazard area as defined by the Federal Emergency Management Agency, and, if so, a flood notification form signed by Borrower and evidence that the flood insurance required by Article V of this Agreement is in place for such Improvements and contents, if applicable, all in form, substance and amount satisfactory to Lender.

"Flood Laws" means the National Flood Insurance Act of 1968, the Flood Disaster Protection Act of 1973, the National Flood Insurance Reform Act of 1994, the Biggert-Waters Flood Insurance Act of 2012, as such statutes may be amended or re-codified from time to time, any substitutions, any regulations promulgated under such Flood laws, and all other Legal Requirements relating to flood insurance.

"Force Majeure Causes" has the meaning in Section 3.12(b).

"Governmental Authority" means the government of the United States of America, any other nation or any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government.

"Guarantor" means Indus Realty Trust, Inc., a Maryland corporation.

J.P.Morgan

"<u>Hard Costs</u>" means costs for work, labor and materials required to construct the Improvements, including those costs identified as "Hard Costs" on the Budget including Tenant Improvement Costs.

"Improvements" means the improvements contemplated by the Plans and Specifications.

"Initial Advance" means the first Advance made in accordance with the terms of this Agreement.

"Initial Maturity Date" means May 7, 2023.

"Inspecting Professional" means CM Advisors, Inc., or any successor selected by Lender.

"Interest Adjustment Date" means the day in each calendar month commencing after the Initial Advance which numerically corresponds to the date of the Initial Advance, provided, however, that (a) if any Interest Adjustment Date would be on a day other than a Business Day, such Interest Adjustment Date shall be the next succeeding Business Day unless such next succeeding Business Day would fall in the next calendar month, in which case such Interest Adjustment Date shall be the next preceding Business Day and (ii) if, for any calendar month, there is no day numerically corresponding to the date of the Initial Advance, Interest Adjustment Date for such calendar month shall be the last Business Day of such month.

"Interest Payment Date" means the fifth (5th) day of each month and the Maturity Date.

"Interest Period" means on any date of determination with respect to any Advance accruing interest at the LIBOR Rate, the period commencing on the then most recent Interest Adjustment Date (provided, however, that with respect to the Initial Advance, the initial Interest Period shall commence on the date of the Initial Advance) and ending on the date immediately prior to the next Interest Adjustment Date.

"Interest Rate" has the meaning in Section 2.04(a).

"IRS" means the United States Internal Revenue Service.

"<u>Lease</u>" means any lease or other agreement for the use and occupancy of all or any portion of the Property, whether now in existence or hereafter arising.

"<u>Leasing Commissions</u>" means a leasing commission payment owing to a leasing agent in connection with an Approved Lease.

"Leasing Costs" means Tenant Improvement Costs and Leasing Commissions.

"<u>Legal Requirements</u>" means any and all judicial decisions, statutes, rulings, directions, rules, regulations, permits, certificates or ordinances of any Governmental Authority in any way applicable to the Loan, Borrower or any of its Affiliates or the Property or any of their respective property, including the ownership, division, use, occupancy, possession, operation, maintenance, alteration, repair or construction thereof.

"Lessee" means a tenant under a Lease.

"Liabilities" has the meaning in Section 7.03(c).

"<u>LIBOR Rate</u>" means for any day the rate for borrowed funds in the interbank market in London in United States dollars for maturities of one month, rounded upwards if necessary to the next 1/16 of 1%, as determined at approximately 11:00 a.m. London time two (2) Business Days prior to such date. Such rate shall be determined

by Lender based upon the London Interbank Offered Rate as administered by the ICE Benchmark Administration (or any other entity that takes over administration of such rate for United States dollars). Any LIBOR Rate less than 0.0% shall be deemed to be 0.0% for purposes of this Agreement, provided that the foregoing floor shall not be applied to determine the LIBOR Rate as used in the definition of "CB Floating Rate".

"Lien" means, with respect to any asset, (a) any mortgage, deed of trust, lien, pledge, hypothecation, encumbrance, charge or security interest in, on or of such asset, (b) the interest of a vendor or a lessor under any conditional sale agreement, capital lease or title retention agreement (or any financing lease having substantially the same economic effect as any of the foregoing) relating to such asset and (c) in the case of securities, any purchase option, call or similar right of a third party with respect to such securities.

"<u>Limited Guaranty</u>" means the Limited Guaranty dated the same date as this Agreement executed by Guarantor in favor of Lender, as amended from time to time, with respect to certain acts of Borrower.

"Line Item Cost Savings" means the portion of any Budget line item which Borrower demonstrates to the satisfaction of Lender is unnecessary for full payment of the cost or expense covered by such Budget line item.

"Loan" means the loan made by Lender pursuant to this Agreement.

"Loan Documents" means this Agreement, including the schedules, exhibits, the Note, the Security Instrument, the Completion Guaranty, the Limited Guaranty, the Environmental Indemnity Agreement, the Architect Subordination Agreement, the Contractor Subordination Agreement, the Engineer Subordination Agreement, and any and all other documents now or hereafter executed by Borrower, Guarantor or any other guarantor of the Obligations or any portion thereof evidencing, guarantying, securing or otherwise pertaining to the Obligations, including any amendments, modifications or supplements or waivers, UCC filings, and any other documents prepared in connection with the other Loan Documents.

"Loan-to-Cost Ratio" means the ratio, expressed as a percentage, of (a) the Commitment to (b) the total amount of Project Costs.

"Loan-to-Value Ratio" means the ratio, expressed as a percentage, of (a) the Commitment to (b) the Appraised Value.

"Maturity Date" means the Initial Maturity Date as such date may be extended pursuant to Section 2.07.

"Minimum Debt Yield" has the meaning given to such term in Section 3.20.

"NOI" means, as of any date of determination, an amount equal to (i) Total Revenues, as of such date of determination, minus (ii) Property Operating Expenses as of such date of determination, in each case construed in accordance with generally accepted accounting principles consistently applied.

"Note" means the Promissory Note made by Borrower to the order of Lender, as amended from time to time.

"Obligations" means all unpaid principal of and accrued and unpaid interest on the Loan, all accrued and unpaid fees and all expenses, reimbursements, indemnities and other indebtedness, liabilities or obligations of Borrower to Lender, or any indemnified party arising under the Loan Documents.

"OFAC" means the Office of Foreign Assets Control of the United States Department of the Treasury.

"Operating Costs" means the costs and expenses for the operation and maintenance of the Project.

"<u>Permitted Encumbrances</u>" means (a) Liens and security interests granted pursuant to the Loan Documents, (b) the items on Schedule B of the Title Policy, (c) customary easements entered into by Borrower in connection with the Project which Lender in its commercially reasonable judgment has determined would have no material adverse effect on the use or value of the Property, and (d) documents required to be recorded by applicable law which have no material adverse effect on the use or value of the Property.

"Permitted Indebtedness" means (a) the Obligations, (b) obligations outstanding under the Construction Contract in such amounts as are normal and reasonable under the circumstances, provided that such debt is not evidenced by a note and is paid when due (or is being disputed in good faith by the Borrower in accordance with the terms and conditions of the Loan Documents), (c) trade debt incurred in the ordinary course of the Project in such amounts as are normal and reasonable under the circumstances, provided that such debt is not evidenced by a note and is paid when due (or is being disputed in good faith by the Borrower in accordance with the terms and conditions of the Loan Documents) and provided in any event that the outstanding principal balance of such debt does not exceed at any one time one percent 1% of the Commitment, and (d) equipment leases entered into in the ordinary course of the Project.

"<u>Permitted Transfer</u>" means any of the following transactions (each an "<u>Indirect Transfer</u>"), if and only if, the Permitted Transfer Requirements are satisfied with respect to such Indirect Transfer:

- (i) any transfer, sale (whether by merger or otherwise), encumbrance or issuance of, publicly traded stock, publicly traded securities or other publicly traded ownership interests in any Person that directly or indirectly owns an interest in Borrower, including a merger, consolidation or other business combination with respect to such Person, or a sale of all or substantially all of such Person's assets (other than the Property); or
- (ii) transfers or encumbrances of direct or indirect interests in Borrower or any direct or indirect owners of Borrower (including transfers of interests to non-"United States persons" within the meaning of the Code).

"<u>Permitted Transfer Requirements</u>" means, collectively, the following requirements with respect to any Indirect Transfer:

- (i) After such Indirect Transfer, Guarantor, directly or indirectly (a) owns more than 51% of the ownership interests in Borrower and (b) Controls Borrower;
- (ii) If the transfer will result in there being a direct or indirect owner of twenty-five percent (25%) or more of Borrower (other than an entity that owns, directly or indirectly, twenty-five percent (25%) or more of Borrower as of the date of this Agreement), Borrower has given Lender at least thirty (30) days prior written notice of such transfer, Borrower has provided to Lender such information about the transferee as Lender may reasonably request and Lender has conducted its customary background, underwriting and internal compliance checks as to the transferees and shall be satisfied with the results in its commercially reasonable judgment;
- (iii) Except for transfers of publicly traded interests in accordance with subparagraph (a) of the definition of Permitted Transfers, such Indirect Transfer: (i) will not result in the representations in Sections 4.01(k) and (l) to not be true, or (ii) will not result in a violation of applicable U.S. Federal law or regulation for Lender to have a loan outstanding to a borrower in which such proposed transferee owns a direct or indirect interest; and
- (iv) No Default exists or will occur upon the consummation of such Indirect Transfer.

"<u>Person</u>" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"<u>Plans and Specifications</u>" means the final plans and specifications for the Improvements accepted by Lender and approved by all applicable Governmental Authorities, as amended from time to time in accordance with this Agreement.

"Prime Rate" means the rate of interest last quoted by The Wall Street Journal as the "Prime Rate" in the U.S. or, if The Wall Street Journal ceases to quote such rate, the highest per annum interest rate published by the Federal Reserve Board in Federal Reserve Statistical Release H.15 (519) (Selected Interest Rates) as the "bank prime loan" rate or, if such rate is no longer quoted therein, any similar rate quoted therein (as determined by Lender) or any similar release by the Federal Reserve Board (as determined by Lender). Each change in the Prime Rate shall be effective from and including the date such change is publicly announced or quoted as being effective.

"Project" means the acquisition of the Land, construction of the Improvements and the leasing and operation of the Improvements.

"<u>Project Costs</u>" means, on the applicable date of determination, all Hard Costs, Soft Costs, Property Acquisition Costs, Leasing Costs and Property Operating Expenses through the date then currently projected for the commencement of Amazon Tenant's obligation to pay base rent under the Amazon Lease.

"Project Documents" means the Development Agreement and any other agreement relating to the ownership, financing, development or operation of the Improvements to which Borrower is a party or beneficiary; provided, however, Project Documents do not include the Loan Documents, the Leases, the Architect's Contract, the Engineer's Contract, the Construction Contract or any subcontract relating to the construction of the Improvements.

"Property Acquisition Costs" means total costs to acquire the Property in the amount of \$7,250,000.

"Property" means the Land and the Improvements.

"Property Operating Expenses" means, as of any date of determination, the sum (without duplication) of the following for the most recently completed 3 month period, annualized: (a) all expenses incurred in connection with the greater of actual expenses or appraised expenses required for the Project including utilities, ordinary repairs and maintenance and insurance premiums, and (b) property management fees for the Project in an amount equal to the greater of the actual property management fees incurred or three percent (3%) of Total Revenues for the applicable period. Property Operating Expenses will be subject to such adjustments as are required in Lender's judgment and will be subject to appropriate adjustment for assessed real estate taxes and seasonal and other adjustments as approved by Lender. Property Operating Expenses excludes debt service charges, depreciation, amortization, and other non-cash expenses.

"Related Parties" means, with respect to any specified Person, such Person's Affiliates and the respective directors, officers, employees, agents and advisors of such Person and such Person's Affiliates.

"Sanctioned Country" means, at any time, a country, region or territory which is itself the subject or target of any Sanctions.

"Sanctioned Person" means, at any time, (a) any Person listed in any Sanctions-related list of designated Persons maintained by OFAC or the U.S. Department of State, (b) any Person operating, organized or resident in a Sanctioned Country, (c) an Affiliate of any such Person described in the foregoing clauses (a) or (b), or (d) any Person otherwise the subject of any Sanctions.

"Sanctions" means all economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by the U.S. government, including those administered by OFAC or the U.S. Department of State.

"Security Instrument" means the Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing dated the same date as this Agreement and executed by Borrower in favor of Lender, as amended from time to time.

"Shell Improvements" means the Improvements other than the required tenant improvements.

"Soft Costs" means those costs for the development, construction, marketing, leasing, operation and maintenance of the Improvements which are not Hard Costs or Tenant Improvement Costs, including the Developer Fee, Leasing Commissions, architectural and engineering fees, consultant and professional fees, marketing fees and expenses, real estate taxes, insurance and bonding costs, interest and financing fees and any other items identified as "Soft Costs" in the Budget.

"Stored Materials" means materials for which Advances will be sought prior to the incorporation into the Improvements.

"Substantially Complete", "Substantially Completed" or "Substantial Completion" means, with respect to the Improvements, that in all material respects, the core and shell of the Improvements (including all structural elements and building systems) and all site improvements have been substantially completed (subject to customary punch-list items) in accordance with the Plans and Specifications and all applicable Legal Requirements, including receipt of a certificate of occupancy (which may be a temporary certificate of occupancy, subject only to conditions as Lender may approve in its commercially reasonable discretion) or a final sign-off and all other construction work required to be performed by Borrower in order for the "commencement date" as defined in the Amazon Lease to occur has been substantially completed in accordance with the terms of the Amazon Lease in all material respects.

"Survey" has the meaning in Exhibit B.

"Swap Agreement" means any agreement, contract or transaction that constitutes a "swap" within the meaning of Section 1a(47) of the Commodity Exchange Act.

"Swap Obligations" of a Person means any and all obligations of such Person, whether absolute or contingent and howsoever and whensoever created, arising, evidenced or acquired (including all renewals, extensions and modifications thereof and substitutions therefor), under (a) any and all Swap Agreements, and (b) any and all cancellations, buy backs, reversals, terminations or assignments of any Swap Agreement transaction.

"<u>Tenant Allowance Payment</u>" means a tenant improvement allowance payment owing to a tenant under the terms of an Approved Lease.

"<u>Tenant Improvement Costs</u>" means Tenant Allowance Payments, and costs required for Borrower to complete tenant improvements on the Property pursuant to its obligations under Approved Leases.

"Tenant Monetary Default" means (i) a default by any Lessee in the payment of scheduled rent under its Lease which continues beyond any applicable grace or cure period, or (ii) the filing of any petition or the commencement of any case or proceeding by or against any Lessee under any Debtor Relief Laws.

"<u>Title Policy</u>" means an ALTA Lender's Policy of Title Insurance in form and substance satisfactory to Lender issued by Chicago Title Insurance Company in the amount of the Loan insuring the Security Instrument as a first priority lien on the Property, containing such endorsements and with such re-insurance as Lender may request, excepting only such items as are acceptable to Lender.

"<u>Total HVCRE Equity</u>" means \$13,024,120, which is an amount equal to the as-is appraised value of any contributed real property and improvements and Cash Equity contributed by Borrower, in each case as approved by Lender. The amount of Total HVCRE Equity shall be reduced by any distribution allowed under Section 3.06.

"Total Revenues" means, as of any date of determination, the sum (without duplication) of the following: for the most recently completed 3 month period, annualized: (a) the actual gross rental revenue received by Borrower under Approved Leases from Lessees in occupancy and paying rent as of such date of determination; (b) the actual expense reimbursement revenue received by Borrower under Approved Leases from Lessees in occupancy and paying rent as of the date of determination; (c) any other actual operating revenue received by Borrower from the Property of a recurring nature. Total Revenues as of any date of determination excludes (1) revenue received under Approved Leases under which a Tenant Monetary Default has occurred and is continuing as of such date of determination, (2) revenue received under Approved Leases that will terminate or expire within 30 days after such date of determination, whether because of the expiration of the Lease term without proper exercise by the Lessee of an available extension option, the exercise by the Lessee of a termination option, or otherwise, (3) revenue received under Approved Leases where the Lessee has not waived in writing any termination options available to such Lessee during the 30 days after such date of determination, (4) lease termination fees, straight-line income, security deposits not yet applied in accordance with the underlying applicable Approved Lease, insurance proceeds and condemnation awards and other extraordinary items of income, and (5) all rental and reimbursement revenues, if any, collected during the applicable calculation period with respect to the space covered by any Approved Lease described in clause (2) above or any prior Lease.

"<u>Transactions</u>" means the execution, delivery and performance by Borrower of this Agreement and the other Loan Documents, the borrowing of the Loan and the use of the proceeds thereof.

"<u>Unmatured Default</u>" means the occurrence of an event which with notice or lapse of time or both would constitute a Default.

ARTICLE II: LOAN TERMS

2.01 Loan.

- (a) <u>Note</u>. Lender agrees, on the terms and conditions in this Agreement, to make the Loan to Borrower in an aggregate amount not to exceed the Commitment. The Loan made by Lender shall be evidenced by the Note.
- (b) <u>Purpose</u>. Borrower agrees to cause the proceeds of each Advance to be applied for Project Costs specified in the applicable Draw Package and approved for disbursement and for no other purposes; provided, however, that Lender shall have no obligation to make any Advance for Project Costs that do not constitute "Allowable Development Costs" as defined and described in the Development Agreement. No portion of any Advance shall be applied by Borrower to reimburse itself for costs previously funded with or credited to Borrower's Equity Requirement or costs funded with any Balancing Deposit.

2.02 Conditions to Closing; Advances.

- (a) <u>Conditions to Closing and Each Advance</u>. Borrower agrees that, in addition to all other conditions herein, the closing of the Loan is conditioned on the fulfillment of the conditions in <u>Exhibit B</u>.
- (b) <u>Funding</u>. Unless otherwise provided in Section 2.02(d), Lender will make Advances to the Borrower on the proposed Funding Date pursuant to the BAQ.
 - (c) Initial Advance and Subsequent Advances.
- (i) <u>Generally</u>. Each Advance shall be subject to the terms of <u>Exhibits B</u> and <u>C</u> as well as the terms of this Section 2.02(c). Borrower shall be entitled to request (and Lender shall be obligated to make) no more than one (1) Advance in any calendar month; provided, however, Lender may make an additional Advance each calendar month to pay interest as provided in Section 2.02(f).
- (ii) Minimum Advance Amounts. Except for Advances to pay interest, any Advance other than the final Advance, if applicable, shall be in an aggregate amount that is not less than \$100,000, unless otherwise approved by Lender.
- (iii) Requests for Advances. To request an Advance, Borrower shall notify Lender of such request by electronic communication as provided in Section 7.01, not later than 11:00 a.m. Dallas, Texas time, ten (10) Business Days before the proposed Borrowing Date of such Advance. Each Borrowing Request shall be irrevocable, shall be in a form approved by Lender and shall specify the following information in compliance with Section 2.02(c)(i) and (ii): (A) the aggregate amount of the requested Advance, and (B) the requested Borrowing Date of such Advance, which shall be a Business Day.

(d) Loan Disbursements/Equity Draws.

(i) <u>Loan Disbursements</u>. Unless a Default or Unmatured Default exists, Advances shall be disbursed by Lender directly to Borrower. Any Advance made by Lender when a Default or Unmatured Default exists shall be disbursed, at Lender's option: (a) directly to Borrower, (b) directly to Contractor, (c) jointly to Borrower and Contractor or any other contractor or subcontractor, (d) directly to Persons supplying labor, materials and services in connection with the Improvements, (e) jointly to Borrower and said Persons, or (f) any combination of the foregoing.

(ii) <u>Borrower's Equity Requirement</u>. Borrower shall submit a Draw Package for each disbursement of Borrower's Equity Requirement to Lender at least ten (10) Business Days in advance of such disbursement as if such disbursement constituted an Advance.

(e) Balancing.

- (i) "In Balance" Determination. Advances shall only be made at such times as the Loan is "in balance". The Loan shall be deemed to be "in balance" only at such times as Lender determines that (A) amounts available for disbursement under the Loan Documents for Project Costs together with (B) available undisbursed Balancing Deposits, will be sufficient to complete and operate the Improvements in accordance with the requirements of this Agreement and pay all Project Costs as and when expected to be incurred through the date then currently projected for the commencement of Amazon Tenant's obligation to pay Lease rent under the Amazon Lease.
- (ii) <u>Balancing Deposits</u>; <u>Security Interest</u>. Within ten (10) days after written notice from Lender that the Loan is not "in balance," and prior to any subsequent Advance, Borrower shall deposit sufficient funds with Lender to bring the Loan "in balance" (in each case, a "<u>Balancing Deposit</u>"). Any amounts so deposited will be held by Lender in an interest bearing account (the "<u>Borrower's Funds Account</u>"). Borrower hereby grants to Lender a first-priority security interest in and lien on the Borrower's Funds Account and all amounts from time to time on deposit, and any proceeds, as security for the Obligations. Any amounts so deposited shall be disbursed for the payment of Project Costs before any additional Advances are made and shall be subject to the same conditions as would be applicable to an Advance. Upon the occurrence, and during the continuance, of a Default, Lender may apply all or any portion of any Balancing Deposit (including accrued interest) to the payment of the Obligations or any Project Costs.
- (f) Advances to Pay Interest. Borrower hereby irrevocably authorizes Lender to make Advances to pay interest when due on the Loan from the portion of the Commitment allocated in the Budget for interest to pay interest (the "Interest Reserve") when due on the Loan. Lender may make such Advances notwithstanding that a Default or Unmatured Default may have occurred. If funds are not available from the Interest Reserve to pay interest when due on the Loan, Borrower shall pay such interest from its own funds. Cash flow from the Project (after payment of Property Operating Expenses) must be used to pay interest on the Loan before any Advances are made to pay interest.

(g) Budget.

- (i) Any changes to the Budget shall require Lender's consent.
- (ii) Line Item Cost Savings from any Budget line item other than for interest on the Loan, shall be reallocated to the contingency line item and shall be available for costs in any other Budget line item.
- (iii) Funds allocated to the contingency line item in the Budget shall be available for Project Costs in the same proportion as the percent of completion of the Project.
- (h) <u>Stored Materials.</u> Borrower shall be entitled to receive Advances for Stored Materials so long as: (A) the amount to be advanced on account thereof does not include the cost of incorporating such materials into the Improvements; (B) the materials are suitably stored on-site (or off-site, as applicable) and insured for the full value against theft, destruction or other casualty under insurance policies designating Lender as loss payee and

additional insured as evidenced by insurance binders or endorsements satisfactory to Lender in its commercially reasonable judgment; (C) immediately upon disbursement of the Advance, Borrower will have absolute title to the Stored Materials as evidenced by appropriate bills of sale and payment receipts; (D) to the extent required by Lender, the Inspecting Professional shall have verified that the materials to be paid for comply with the Plans and Specifications and are of suitable quality for ultimate incorporation into the Improvements and are free from any apparent defect; (E) to the extent required by Lender, any off-site Stored Materials and components shall be physically segregated from the other assets of the vendor or placed in a bonded warehouse or similarly secured facility, and (F) Lender shall have a perfected security interest in the Stored Materials. Notwithstanding the foregoing, in no event shall Advances for off-site Stored Materials exceed an aggregate of Three Million and No/100 Dollars (\$3,000,000.00) at any one time of which up to One Million and No/100 Dollars (\$1,000,000.00) may be for deposits for materials to be incorporated into the Project

- (i) Hard Costs; Soft Costs; Leasing Costs.
- (i) Advances for Hard Costs and Soft Costs shall be made in accordance with the Budget and the provisions in the Construction Contract for the withholding and payment of retainage. Lender shall not be required to disburse more than the amount shown in the Budget for any Project Cost. The line item in the Budget identified as "Developer's Fee" shall be available for disbursement to Borrower at any time in the same proportion as the then current percentage completion of the Improvements.
- (ii) Advances for Tenant Improvement Costs/Tenant Allowance Payments shall be limited to the amount set forth in the Budget and Advances for Leasing Commissions shall be limited to \$1,000,000 with respect to the Amazon Lease, or such other amounts in each case as Lender may approve in writing in connection with Lender's specific approval of any other Lease.
- (j) <u>Final Construction Advance</u>; <u>Release of Retainage</u>. Advances for the payment of final construction costs and release of retainage to the Contractor shall not be made until Lender has received and approved, in its commercially reasonable judgment, all of the following:
 - (i) evidence that Substantial Completion has occurred;
- (ii) such title insurance endorsements as Lender may reasonably require including, without limitation, endorsements to the Title Policy insuring the Security Instrument as a first lien upon the Property, excepting only such items as shall be permitted under the Loan Documents, and insuring over all mechanics' and material suppliers' liens arising (or which may arise) from the completion of the Improvements;
- (iii) if requested by Lender, an ALTA as-built survey or other satisfactory evidence (which includes an appropriate professional seal) showing that (A) the Shell Improvements do not encroach on any easement or public or private right of way, (B) the Shell Improvements have been constructed within the boundaries of the Property, (C) the Shell Improvements have been constructed within the setback lines as required by applicable zoning ordinances and do not encroach upon any other lot or property;
- (iv) if requested by Lender, "as-built" Plans and Specifications of the Shell Improvements, showing the final specifications of all Improvements and that in all material respects the Improvements have been built substantially in accordance with the Plans and Specifications;

- (v) if requested by Lender, executed AIA Form G704 (Certificate of Substantial Completion), executed AIA Form G706 (Contractor's Affidavit of Payments of Debts and claims), AIA Form G706A (Contractor's Affidavit of Release of liens) and if Contractor has a payment and performance bond, AIA Form G707 (Consent of Surety of Final payment);
- (vi) if requested by Lender, a warranty book, together with all guaranties and maintenance agreements on all Improvements; and
- (vii) if requested by Lender and if the same is customary in connection with the completion of construction where the Property is located, a notice of completion duly recorded in the official records of the county where the Property is located.

Notwithstanding the foregoing, if a particular subcontractor completes its scope of work prior to Substantial Completion, the retainage held with respect to such subcontractor will be advanced to Borrower at Borrower's request, subject to (i) satisfaction of each of the conditions for Advances (other than Advances for retainage) and (ii) receipt by Lender of a conditional lien waiver, followed by delivery to Lender of a final unconditional lien waiver, from the applicable subcontractor within sixty (60) days.

2.03 **Fees**.

- (a) Loan Fee. Borrower will pay Lender a loan fee of \$127,800.00 on the Closing Date.
- (b) <u>Fees Generally</u>. All fees payable under this Agreement will be paid in immediately available funds and are nonrefundable.

2.04 Interest.

(a) Interest Rate.

- (i) <u>Initial Advance</u>; <u>Subsequent Advances</u>. The Initial Advance and any subsequent Advances made on an Interest Adjustment Date shall bear interest at a per annum interest rate equal to the LIBOR Rate plus the Applicable Margin. Any subsequent Advances made on any date other than an Interest Adjustment Date shall bear interest at a per annum interest rate equal to the Daily Floating LIBOR Rate plus the Applicable Margin as the same shall be in effect on each day from the date of such Advance through and including the date immediately prior to the next Interest Adjustment Date.
- (ii) <u>Monthly Reset of Interest Rate</u>. Commencing on the Interest Adjustment Date of the first calendar month after the Initial Advance, and continuing thereafter on each Interest Adjustment Date, the interest rate applicable to the then outstanding principal balance of the Loan shall be reset to the Interest Rate as then in effect.
- (b) <u>Default Rate</u>. To the extent permitted under applicable law, upon the occurrence of a Default, and after maturity, the outstanding principal balance of the Loan will bear interest, before and after judgment, at a rate per annum equal to five percent (5%) plus the rate otherwise applicable to the outstanding principal balance of the Loan or, at the option of Lender, five percent (5%) plus the CB Floating Rate plus the Applicable Margin (as applicable, "Default Rate").
- (c) <u>Payment of Accrued Interest</u>. Accrued interest on the outstanding principal balance of the Loan is payable in arrears on each Interest Payment Date for the immediately preceding calendar month; provided (i)

interest accrued pursuant to Section 2.04(b) above will be payable on demand, (ii) in the event of any repayment or prepayment of any principal of the Loan, accrued interest on the principal amount repaid or prepaid will be payable on the date of such repayment or prepayment and (iii) in the event of any conversion of the rate of interest on the Loan to the rate based on the CB Floating Rate, accrued interest on the Loan will be due and payable on the effective date of such conversion.

(d) <u>Computation of Interest</u>. All interest hereunder will be computed on the basis of a year of 360 days and will be payable for the actual number of days elapsed (including the first day but excluding the last day). The applicable interest rate for any day will be determined by Lender in accordance with the terms hereof.

2.05 Repayment of Loan.

(a) Payments Generally. Borrower will make each payment required to be made by it under this Agreement prior to 11:00 a.m., Dallas, Texas time, on the date when due, in immediately available funds, without setoff or counterclaim. Any amounts received after such time on any date may, in the discretion of Lender be deemed to have been received on the next succeeding Business Day. All such payments will be made to Lender at 700 N. Pearl Street, Floor 13, Dallas, TX 75201-2824. If any payment under this Agreement becomes due on a day that is not a Business Day, such payment will be made on the next succeeding Business Day, and, in the case of any payment accruing interest, interest will continue to accrue during the extension period. All payments hereunder will be made in U.S. dollars.

(b) Reserved.

- (c) <u>Repayment at Maturity</u>. Borrower hereby unconditionally promises to pay to Lender on the Maturity Date all unpaid Obligations then due.
- (d) <u>Prepayment</u>. Borrower may at any time and from time to time to prepay the Loan in whole or in part, subject to prior notice in accordance with this Section 2.05. Borrower will notify Lender by electronic communication as provided in Section 7.01 of any prepayment not later than 11:00 a.m., Dallas, Texas time, three (3) Business Days before the date of prepayment. Each such notice will be irrevocable and must specify the prepayment date and the principal amount of the Loan to be prepaid. Each partial prepayment of the Loan will be in an amount not less than \$1,000,000.00.
- (e) <u>Late Fee</u>. If any payment is not received by Lender within ten (10) days after its due date, Lender may assess and Borrower agrees to pay a late fee equal to five percent (5%) of the past due amount except for the unpaid principal amount of the Loan on the Maturity Date. Borrower will pay the late fee upon demand by Lender.

2.06 Alternate Rate of Interest.

(a) If (i) Lender determines that adequate and reasonable means do not exist for ascertaining the LIBOR Rate for any Interest Period or the Daily Floating LIBOR Rate for any day (including because the LIBOR Rate or the Daily Floating LIBOR Rate is not available or published on a current basis), or (ii) Lender determines that the LIBOR Rate for any Interest Period or the Daily Floating LIBOR Rate for any day will not adequately and fairly reflect the cost to Lender of making or maintaining the Loan, then Lender shall give notice to Borrower by electronic communication as provided in Section 7.01 as promptly as practicable and, until Lender notifies Borrower that the circumstances giving rise to such notice no longer exist, the Loan shall bear interest at the CB Floating Rate plus the Applicable Margin.

- (b) If at any time Lender determines that (i) the circumstances in clause (a)(i) have arisen and such circumstances are unlikely to be temporary or (ii) the supervisor for the administrator of the LIBOR Rate or the Daily Floating LIBOR Rate or a Governmental Authority having jurisdiction over Lender has made a public statement that the LIBOR Rate or the Daily Floating LIBOR Rate, as applicable, (x) has ceased, or will cease, to be provided permanently or indefinitely (provided that there is no successor administrator that will continue to provide the LIBOR Rate or the Daily Floating LIBOR Rate), or (y) is no longer representative, then Lender and Borrower may amend this Agreement to (1) establish an alternate rate of interest to the LIBOR Rate or the Daily Floating LIBOR Rate, which alternate rate may include a mathematical adjustment that gives due consideration to the then prevailing market convention or to recommendation by a relevant Governmental Authority (as determined by Lender) for determining a spread adjustment for the replacement of the LIBOR Rate or the Daily Floating LIBOR Rate, (2) reflect any technical, administrative or operational changes that Lender requires in its reasonable discretion to implement and administer such alternate rate of interest and (3) make any other related changes to this Agreement as may be applicable. Until an alternate rate of interest shall be determined in accordance with this clause (b) (but, in the case of the circumstances described in clause (ii) of the first sentence of this Section 2.06(b), only to the extent the LIBOR Rate for such Interest Period or the Daily Floating LIBOR Rate, as applicable, is not available or published at such time on a current basis), the Loan shall bear interest at the CB Floating Rate plus the Applicable Margin. In the event that any alternate rate of interest established pursuant to this Section 2.06(b) shall be less than zero percent (0%), such rate shall be deemed to be zero percent (0%) for the purposes of this Agreement.
- (c) All determinations by Lender under this Section 2.06 shall be conclusive and binding absent manifest error.

2.07 Extension Option.

- (a) <u>Extension Option</u>. Borrower has the option to extend the Initial Maturity Date to the one-year anniversary of the Initial Maturity Date provided the following conditions are satisfied with respect to the applicable Extension:
- (i) Borrower has provided Lender with written notice of Borrower's request to exercise an extension option at least 45 days but not more than 120 days prior to the Maturity Date in effect prior to such Extension;
- (ii) No Default or Unmatured Default has occurred and is continuing on the Maturity Date in effect prior to such Extension;
- (iii) All representations and warranties made under this Agreement or under any other Loan Documents are true and correct in all material respects as of the Maturity Date in effect prior to such Extension, except to the extent any such representation and warranty is made as of a specified date, in which case such representation and warranty shall have been true and correct as of such specified date;
- (iv) Lender has received satisfactory documentation evidencing such Extension executed by Borrower and consented to by Guarantor; Lender has received any title insurance endorsements reasonably requested by Lender in connection with such Extension; and Borrower has paid all of Lender's costs and expenses, including reasonable attorneys' fees, in connection with such Extension;
- (v) Borrower is in compliance with the Flood Insurance Requirements;

- (vi) Borrower has paid to Lender an extension fee in an amount equal to two-tenths of one percent (0.20%) of the Commitment then in effect;
- (vii) The Improvements are Substantially Complete;
- (viii) No material adverse change has occurred in the financial or other condition of Borrower, Guarantor, Amazon Tenant, Amazon Parent, the Project or the Property;
- (ix) (A) Amazon Tenant is in occupancy under the Amazon Lease, is paying rent as of the Initial Maturity Date and has delivered to Lender a tenant estoppel certificate in form and substance reasonably acceptable to Lender dated no earlier than thirty (30) days prior to the Initial Maturity Date, and (B) no default or event of default has occurred and is continuing under the Amazon Lease or the Amazon Guaranty.
 - (b) Reserved.
- (c) <u>Cancellation of Commitment</u>. Any undisbursed portion of the Commitment shall be permanently cancelled on the Initial Maturity Date and the Commitment automatically reduced to the outstanding principal balance of the Loan.

ARTICLE III: GENERAL COVENANTS; CONSTRUCTION

3.01 Liens, Taxes, and Governmental Claims; Right to Contest.

- (a) <u>Liens</u>. Borrower will promptly pay, satisfy and obtain the release of all claims or Liens on any part of the Property (other than the Permitted Encumbrances), and all costs, charges, interest arising in connection therewith, and, upon demand, will give Lender evidence satisfactory to Lender of the payment, satisfaction or release of such claims or Liens.
- (b) <u>Taxes</u>. Borrower agrees to pay or cause to be paid, prior to the date they would become delinquent if not paid, any and all taxes, assessments and governmental charges levied upon or assessed or charged against the Property, including all water and sewer taxes, assessments and other charges, fines, impositions and rents, if any. If requested by Lender, Borrower will give to Lender a receipt or receipts, or certified copies, evidencing every such payment by Borrower, not later than forty-five (45) days after such payment is made.
- (c) Contest. Borrower may dispute or contest and will not be required to pay any taxes, claims or governmental charges, or claims or Liens as required by Sections 3.01(a) or (b), so long as (i) Borrower provides prompt written notice of such dispute or contest to Lender, (ii) Borrower diligently prosecutes such dispute or contest to a prompt determination in a manner not prejudicial to Lender and promptly pays all amounts ultimately determined to be owing, and (iii) Borrower provides security or other forms of assurances for the payment of such tax, assessment or governmental charge, or claim, or Lien (together with any related interest and penalties) in an amount and in form and substance satisfactory to Lender in its commercially reasonable judgment. If Borrower shall fail to pay any such amounts ultimately determined to be owing or to proceed diligently to prosecute such dispute or contest as provided herein, then, upon the expiration of ten (10) days after written notice to Borrower by Lender, Lender may, but will not be obligated to, discharge the same, and the costs shall be reimbursed by Borrower to Lender on demand. Lender will be subrogated to all rights, equities and liens discharged by any such expenditure to the fullest extent permitted by law.

3.02 **Leases**.

Except for Approved Leases that do not require Lender's approval, all commercial Leases will be subject to the prior written approval of Lender, which approval may be conditioned upon Lender's receipt of a tenant estoppel certificate and a subordination, non-disturbance and attornment agreement in form acceptable to Lender.

- (a) Affirmative Covenants. Borrower will (i) duly and punctually observe and perform in all material respects its obligations as landlord under the Leases, (ii) give prompt written notice to Lender of any failure on the part of Borrower to observe and perform the same or of any written claim made by any Lessee of any such failure by Borrower, (iii) enforce the performance of every material obligation in the Leases to be performed by any Lessee or any lease guarantor, (iv) appear in and defend any action or proceeding arising under, occurring out of or in any manner connected with the Leases or the obligations, duties or liabilities of Borrower and any Lessee thereunder, do so in the name and on behalf of Lender upon request by Lender, but at the expense of Borrower, and pay all costs and expenses of Lender, including reasonable attorneys' fees and disbursements, in any action or proceeding in which Lender may appear, and (v) cause each Lessee under a Lease to execute and deliver to Lender a tenant estoppel certificate and a subordination, non-disturbance and attornment agreement, each in form and substance acceptable to Lender in its commercially reasonable judgment, promptly upon execution of such Lease (with such revisions as are acceptable to Lender in each case).
- (b) Negative Covenants. Unless Borrower first obtains the written consent of Lender, Borrower will not (i) enter into any Lease that is not an Approved Lease, (ii) cancel, terminate or consent to any surrender of any Lease unless Borrower is compelled to do so under court order, (iii) commence any action of ejectment or any summary proceedings for dispossession of any Lessee under any Lease or exercise any right of recapture provided in any Lease, (iv) modify or alter the terms of any Lease in any material respect, (v) renew or extend the term of any Lease, except as required under the terms of such Lease, (vi) except as required under the terms of any Lease, consent to any subletting of the premises under such Lease, to any assignment of any Lease by the Lessee, or to any assignment of or further subletting of any sublease of any Lease, (vii) receive or collect any rents from any Lessee for a period of more than one month in advance or (viii) further pledge, transfer, mortgage or otherwise encumber or assign future payments of rents under Leases.
- Special Purpose Entity Requirements. Without limitation of any other provisions of this Agreement or any other Loan Document, Borrower hereby represents, warrants, covenants and agrees that it has not and will not: (i) engage in any business or activity other than the acquisition, development, construction, ownership, leasing, operation and maintenance of the Property, and incidental activities; (ii) acquire or own any material asset other than the Property and such incidental personal property as may be necessary for the Project; (iii) merge into or consolidate with any Person or wind-up, dissolve, terminate, divide (whether pursuant to a "plan of division" or other similar arrangement) or liquidate in whole or in part, transfer or otherwise dispose of all or substantially all of its assets or change its legal structure, without in each case the prior written consent of Lender; (iv) commingle its assets with the assets of any other Person; (v) incur any indebtedness, whether direct or contingent, other than Permitted Indebtedness; (vi) allow any Person to pay Borrower's debts and liabilities (except pursuant to the Loan Documents), or fail to pay its debts and liabilities solely from its own assets; (vii) fail to maintain its records, books of account and bank accounts separate and apart from those of any other Person, or fail to prepare and maintain its own financial statements in accordance with generally accepted accounting principles consistently applied; (viii) enter into any contract or agreement with any shareholder, partner, member, principal or Affiliate of Borrower, Guarantor or any shareholder, partner, member, principal or Affiliate thereof, except upon commercially reasonable terms and conditions; or (ix) fail

either to hold itself out to the public as a legal entity separate and distinct from any Person or to conduct its business solely in its own name.

- 3.04 Appraisals. Lender will have the right to order new Appraisals of the Property from time to time. Borrower agrees upon demand to pay to Lender the cost and expense for each Appraisal and a fee for Lender's review of such Appraisal, if such Appraisal is (a) required in connection with the closing of the Loan, (b) obtained in connection with any modification of the Loan, (c) required by any Legal Requirement, or (d) ordered in connection with Borrower's failure to comply with any financial covenant requirement in this Agreement or after the occurrence of a Default.
- 3.05 Operating and Reserve Accounts. Borrower will maintain all operating and reserve accounts for the Project with Lender, and Borrower hereby grants to Lender a security interest in such accounts to secure the Obligations. Borrower will maintain all operating and reserve accounts for the Project free and clear of any Lien, security interest, encumbrance or restriction, except a security interest in favor of Lender. Borrower hereby grants to Lender a security interest in such accounts to secure the Obligations, and Borrower will provide to Lender a control agreement and any other assurance reasonably requested by Lender to perfect such security interest.
- 3.06 **Prohibited Distributions**. Until the Loan has been paid in full and the Commitment terminated, Borrower shall not make any distributions or dividends from the Project or Borrower; provided that distributions or dividends will be allowed so long as (a) no Default or Unmatured Default has occurred and is continuing or will result from such proposed distribution or dividend, (b) the Loan remains "in balance" pursuant to Section 2.02(e) and will do so following such proposed distribution or dividend, (c) Substantial Completion has occurred, and (d) the Debt Service Coverage Ratio as of the most recent calendar quarter is not less than 1.10: 1.00, in each case as determined by Lender.
- 3.07 <u>USA Patriot Act</u>. Lender hereby notifies Borrower that (a) pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Act"), it is required to obtain, verify and record information that identifies Borrower, which information includes the name and address of Borrower and other information that will allow Lender to identify Borrower in accordance with the Act. Borrower will, promptly following a request by Lender, provide all documentation and other information that Lender requests in order to comply with its ongoing obligations under the "know your customer" and anti-money laundering rules and regulations, including the Act and the Beneficial Ownership Regulation. Borrower may not be or become subject at any time to any law, regulation, or list of any Governmental Authority (including the U.S. Office of Foreign Asset Control list) that prohibits or limits Lender from making any advance or extension of credit to Borrower or from otherwise conducting business with Borrower.

3.08 Financial Information and Other Deliveries.

(a) Borrower.

(i) Within sixty (60) days after the end of each of Borrower's fiscal quarters, Borrower will deliver to Lender by electronic communication as provided in Section 7.01 (A) an operating statement (showing actual to budgeted results) and a lease status report (including a rent roll) for the Property, each dated as of the last day of such fiscal quarter, and (B) a balance sheet and statement of operations for Borrower, each dated as of the last day of such fiscal quarter, in form and substance reasonably satisfactory to Lender and certified by an authorized officer of Borrower; provided, however, Borrower shall not be required to deliver operating

statements until commencement of operations in the Improvements or lease status reports (including rent rolls) until commencement of leasing activities.

- (ii) Subject to Section 3.08(a)(i), within ninety (90) days after the end of each of Borrower's fiscal years, Borrower will deliver to Lender by electronic communication as provided in Section 7.01 a balance sheet, statement of operations and statement of cash flow for Borrower, each dated as of the last day of such fiscal year, in form and substance reasonably satisfactory to Lender and certified by an authorized officer of Borrower.
- (iii) Concurrently with the delivery of the information referred to in Section 3.08(a)(i) and (ii), Borrower will deliver to Lender by electronic communication as provided in Section 7.01, a compliance certificate substantially in the form of Exhibit E.
- (iv) Promptly following any officer of Borrower obtaining actual knowledge of any of the following, Borrower will deliver to Lender written notice of (x) the occurrence of any Default or Unmatured Default or the occurrence of an event which would make any representation or warranty contained herein untrue or misleading in any material respect as of the date of such event, (y) any change in the information provided in the Beneficial Ownership Certification that would result in a change to the list of beneficial owners identified in such certification, or (z) the occurrence of any Force Majeure Cause.
- (v) Promptly following any officer of Borrower obtaining actual knowledge of any of the following, Borrower will deliver to Lender written notice of the commencement of any actions, suits or other legal proceedings against or affecting Borrower, Guarantor or the Project.
- (v) Borrower shall deliver to Lender such other information and materials with respect to Borrower, Guarantor, the Project or the Property or compliance with the terms of the Loan Documents, as Lender may reasonably request.

(b) Guarantor.

- (i) Within sixty (60) days after the end of each of Guarantor's fiscal quarters, Borrower will deliver to Lender by electronic communication as provided in Section 7.01 a balance sheet, statement of operations, statement of cash flow for Guarantor, each dated as of the last day of such fiscal quarter, in form and substance reasonably satisfactory to Lender and certified by the chief financial officer of Guarantor;
- (ii) Within ninety (90) days after the end of each of Guarantor's fiscal years, Borrower will deliver to Lender by electronic communication as provided in Section 7.01 a balance sheet, statement of operations and statement of cash flow for Guarantor, each dated as of the last day of such fiscal year, in form and substance reasonably satisfactory to Lender and audited by an independent auditor acceptable to Lender, which acceptance shall not be unreasonably withheld, conditioned or delated.
- (iii) Concurrently with the delivery of the information referred to in Section 3.08(b)(i) and (ii), Borrower will deliver to Lender by electronic communication as provided in Section 7.01, a compliance certificate substantially in the form of Exhibit F.
- (iv) Borrower will promptly deliver to Lender such other information and materials with respect to Guarantor as Lender reasonably requests.

3.09 **ERISA**. At any time during which this Loan is in effect, the representations and warranties in Sections 4.01(i) and (j) will continue to be accurate, and Borrower will not take any action to jeopardize the accuracy of such representations and warranties.

3.10 Compliance with Legal Requirements; Right to Contest.

- (a) Borrower will comply with all Legal Requirements applicable to it or its property (including the Property), except where the failure to do so, individually or in the aggregate, could not reasonably be expected to result in a material adverse effect.
- (b) Notwithstanding any provision of this Agreement or any of the other Loan Documents to the contrary, no Default or Unmatured Default will occur under this Agreement solely as a result of the failure of Borrower, the Project or the Property to comply with any Legal Requirement so long as the following conditions are satisfied: (i) Borrower is contesting the applicability of such Legal Requirement to Borrower, the Project or the Property in good faith and has so notified Lender; (ii) the contest will not materially impair the ability to ultimately comply with the contested Legal Requirement should the contest not be successful and the conduct of the contest will not materially impair Borrower's ability to perform Borrower's obligations under the Loan Documents; (iii) Borrower demonstrates to Lender's reasonable satisfaction that Borrower has the financial capability to undertake and pay for such contest and any corrective or remedial action likely to be necessary; and (iv) Lender is not at risk for any material liability due to Borrower's non-compliance with such Legal Requirement.
- 3.11 <u>Use of Proceeds</u>. Borrower will not request any Transaction and none of Borrower, Guarantor, their respective directors and officers, or any Person participating in any capacity in any Transaction will use the proceeds of the Transactions (a) in furtherance of an offer, payment, promise to pay, or authorization of the payment or giving of money, or anything else of value, to any Person in violation of any Anti-Corruption Laws, (b) for the purpose of funding, financing or facilitating any activities, business or transaction of or with any Sanctioned Person, or in any Sanctioned Country, except to the extent permitted for a Person required to comply with Sanctions, or (c) in any manner that would result in the violation of any Sanctions applicable to any party hereto.

3.12 Acceptance of Construction Documents; Completion of Construction.

- (a) <u>Acceptance of Construction Documents</u>. Lender's acceptance of the Plans and Specifications, the Budget, the Architect's Contract, the Construction Contract, the Engineer's Contract, subcontracts, bonds and other construction documents (including Lender's acceptance of any modifications to, and any Person providing work, labor or services under such agreements) shall not be deemed, and Lender hereby expressly disclaims, any representation or warranty, express or implied, that the Improvements if constructed in accordance with the Plans and Specifications will be structurally sound or have any particular value.
- (b) Completion of Construction. Borrower shall cause the Improvements to be Substantially Completed on or before the Completion Date; provided, however, that the time within which the Improvements must be Substantially Completed may be extended for a period equal to the period of delays caused by any of the following which results in a delay directly affecting construction of the Improvements, but only to the extent beyond the reasonable control of Borrower, provided, however, that the lack of Borrower's own funds shall not be deemed a cause beyond the reasonable control of Borrower: (a) acts of God, including severe weather conditions, earthquakes and flooding, (b) fire or explosion, (c) industry-wide strikes, lockouts. shortages or unavailability of labor or materials, (d) riots or civil disturbance, (e) acts of war or terrorism, (f) orders of governments or civil or military authorities; and (g) quarantine or other restrictions on operations or activities as

a result of pandemic, epidemic, disease, contagion or other health conditions (collectively, "Force Majeure Causes"), but in no event may the time for Substantial Completion of the Improvements extend for more than one hundred twenty (120) days beyond the Completion Date.

3.13 Construction Progress.

- (a) <u>Material Change Orders</u>. There shall be no material change in the Plans and Specifications without the prior consent of Lender. A material change shall be any change (whether such change increases or decreases the total cost of the Improvements), which (i) involves a cost of more than (A) for any single item, Five Hundred Thousand Dollars (\$500,000) or (B) for all such items (without netting cost increases against cost savings), Two Million Five Hundred Thousand Dollars (\$2,500,000), (ii) impairs the structural integrity or the configuration of the Improvements, (iii) substantially changes the architectural appearance of the Improvements or (iv) results in a violation of any applicable Legal Requirement. All changes in the Plans and Specifications must, prior to being effective, be duly approved by applicable bond sureties (if required in order to maintain the effectiveness of the bonds), and all other Persons reasonably required by Lender. Lender's consent to any change in the Plans and Specifications may be conditioned upon Borrower's compliance with a demand for a Balancing Deposit pursuant to Section 2.02(e).
- (b) <u>Foundation Survey and Endorsement</u>. To the extent required by Lender, within thirty (30) days after completion of the construction of the foundation of the Improvements, and as a condition precedent to any further Advances, Borrower shall deliver or cause to be delivered to Lender an update to the Survey showing that the location of such foundation and an endorsement to the Title Policy insuring that such foundation is within the boundary lines of the Property, does not violate any applicable covenants, conditions, restrictions or agreements affecting the Property, and does not encroach upon any easements or rights of way affecting the Property or any portion thereof.
- (c) <u>Title Policy Endorsements</u>. Within thirty (30) days after Substantial Completion of the Improvements, Lender shall receive a date down to the Title Policy.
- 3.14 <u>Purchase of Materials Under Conditional Sales Contract</u>. No materials, equipment, fixtures or any other part of the Property shall be purchased or installed under any security agreement or other arrangements wherein the seller reserves or purports to reserve the right to remove or to repossess any such items or to consider them personal property after their incorporation in the work of construction, unless consented to by Lender in writing.

3.15 Inspection; Inspecting Professional.

- (a) <u>Inspections</u>. Lender, through its officers, agents and employees, shall have the right at all reasonable times, on reasonable prior notice (i) to enter upon the Property and inspect the work of construction; and (ii) to examine the books, records, accounting data and other documents pertaining to the Project or the Property. Borrower will cooperate with Lender and its representatives and consultants.
- (b) <u>Inspecting Professional</u>. Lender may, at its option, require a monthly inspection of the Property by the Inspecting Professional during the construction of the Improvements and any tenant improvement work (as permitted under the terms of the applicable Lease) at Borrower's expense. Without limitation of the provisions of Section 3.15(a), Borrower shall provide the Inspecting Professional with copies of any reports and supporting materials, including testing reports received by Borrower, Architect, Contractor or any engineer with respect to the Property or the Project promptly upon Borrower's receipt thereof.

- (c) Exculpation. It is expressly understood and agreed that Lender is under no duty to supervise or to inspect the work of construction. Any inspections of the Improvements made by or on behalf of Lender are for the sole benefit of Lender and neither Borrower nor any other Person will have any right to rely thereon. Failure to inspect the work or any part thereof shall not constitute a waiver of any of Lender's rights. Inspection not followed by notice of Default shall not constitute a waiver of any Default then existing; nor shall it constitute an acknowledgment that there has been or will be compliance with the Plans and Specifications or applicable Legal Requirements or that the construction is free from defective materials or workmanship. It is further understood and agreed that any consents or approvals of Lender are for the sole purpose of protecting the interests of Lender under the Loan Documents and neither Borrower nor any other Person shall have any right to receive and/or rely on such approvals for any purposes.
- (d) <u>Authority of Inspecting Professional</u>. (i) Inspecting Professional shall not have any power or authority to make any decision or to give any approval or consent or to do any other act or thing which is binding upon Lender, (ii) Lender will not be bound by any opinion or advice of Inspecting Professional, and (iii) Lender reserves the right in its sole and absolute discretion to replace Inspecting Professional with another inspecting professional at any time and without prior notice to or approval by Borrower. Borrower shall have no right to receive copies of any written reports by Inspecting Professional, but in the event Lender does make such information or any portions available to Borrower, Borrower shall rely thereon at its own risk.
- 3.16 Right to Post Signs; Publicity. On Lender's request, Borrower will allow Lender to share signage on the Property for the purpose of identifying Lender as the bank for the construction financing for the Improvements. The form of such signage shall be subject to all Legal Requirements and subject to the prior approval of Borrower, such approval not to be unreasonably withheld or delayed. Borrower shall permit Lender to publicize its involvement in the construction financing for the Property or the Project with Borrower's prior written approval (not to be unreasonably withheld or delayed).
- 3.17 <u>Construction Performance; No Advance Payments; Enforcement</u>. Borrower shall perform and observe in a timely manner each and all material covenants, conditions, obligations and agreements on the part of Borrower to be performed or observed under the Construction Contract, the Engineer's Contract, the Architect's Contract, and the Development Agreement. Borrower shall not waive, excuse, release or discharge Contractor, Engineer or Architect from any material covenants, conditions, obligations or agreements to be performed or observed by Contractor, Engineer or Architect, but shall, at its sole cost and expense, enforce and secure the performance of all material covenants, conditions, obligations and agreements to be performed or observed by Contractor, Engineer or Architect or related to the Construction Contract, the Engineer's Contract or the Architect's Contract.
- 3.18 **No Amendments of Construction Agreements**. Borrower shall not alter, amend or change in any material respect, or terminate or cancel, the Construction Contract, the Engineer's Contract, the Architect's Contract, or the Development Agreement without obtaining Lender's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed.
- 3.19 **No Conveyance of Construction Agreements**. Borrower shall not convey, transfer or assign the Construction Contract, the Engineer's Contract, the Architect's Contract, the Development Agreement or any right, title or interest therein and Borrower shall not execute any other instrument which might prevent or limit Lender from operating under the terms and provisions of the assignment contemplated by the Contractor Subordination Agreement, Engineer Subordination Agreement, and the Architect Subordination Agreement.

3.20 <u>Remargin Obligation</u>. If the base rent payable under the Amazon Lease is reduced in connection with Amazon's audit rights with respect to construction costs related to the Project under the Amazon Lease, and after giving effect to such base rent reduction, the Project fails to generate a Debt Yield of at least nine percent (9.0%) (the "Minimum Debt Yield"), then within ten (10) days of Lender's demand, Borrower shall prepay principal of the Loan by the amount necessary to satisfy the Minimum Debt Yield. Any such prepayment will permanently reduce the Commitment.

ARTICLE IV: REPRESENTATIONS AND WARRANTIES

- 4.01 <u>Representations and Warranties</u>. As a material inducement to Lender to enter into this Agreement, and as an express condition to each Advance made, Borrower hereby represents and warrants, as follows:
- (a) Existence; Power and Authority. Borrower is a limited liability company duly formed and validly existing under the laws of the State of North Carolina, with requisite power and authority to (i) incur the Obligations, and (ii) execute, deliver and perform this Agreement and the other Loan Documents to which it is a party.
- (b) <u>Authorization; No Conflict</u>. Borrower's execution and delivery to Lender of this Agreement and the other Loan Documents and the full and complete performance of the provisions thereof (i) do not violate Borrower's organizational documents; (ii) have been duly authorized by all requisite actions; (iii) do not require the approval or consent of any Governmental Authority having jurisdiction over Borrower or the Property except for consents or approvals that have been obtained or shall be obtained as construction of the Improvements progresses and Borrower has no knowledge of information suggesting that any approvals, licenses and/or permits which have not already been obtained but which will be necessary as construction of the Improvements progresses will not be obtained in a timely manner sufficient to allow Substantial Completion of the Improvements on or before the Completion Date; and (iv) will not result in any breach of, or constitute a default under, or result in the creation of any Lien (other than those contained in any of the Loan Documents) upon any property or assets of Borrower under any indenture, mortgage, deed of trust, credit agreement or other instrument or agreement to which Borrower is a party or by which Borrower or any of the Property is bound.
- (c) <u>Title</u>. Borrower is the sole legal and beneficial owner of the Property free and clear of all Liens other than the Permitted Encumbrances.
 - (d) Financial Statements and Disclosures.
- (i) Any and all balance sheets, statements of income or loss, and financial statements furnished to Lender with respect to Borrower or Guarantor are true and correct in all material respects as of the dates thereof, and fully and accurately present the financial condition as of the dates thereof, and no material adverse change has occurred or is likely to occur in the financial condition reflected therein since the date of the most recent financial statements furnished to Lender.
- (ii) As of the Closing Date, to the best knowledge of Borrower, the information included in the Beneficial Ownership Certification provided on or prior to the Closing Date to Lender in connection with this Agreement is true and correct.
- (e) <u>Litigation</u>. Except for litigation specifically described on <u>Exhibit D</u>, there is no action, suit or other legal proceeding pending, or to the actual knowledge of Borrower, threatened, against Borrower, Guarantor or the Property which (i) if adversely determined would materially and adversely affect the ability of Borrower or

Guarantor to perform its obligations under the Loan Documents or Project Documents or would have a material adverse effect on the use or value of the Property, or (ii) challenge the validity or enforceability of the Loan Documents or the priority of the Liens created thereby.

- (f) <u>Legal Compliance</u>. The Plans and Specifications have been approved by all applicable Governmental Authorities, and upon completion in accordance with the Plans and Specifications, the use and occupancy of the Property will comply in all material respects with all applicable Legal Requirements. Neither the zoning nor any other right to construct, use or operate the Property is to any extent dependent upon or related to any real estate other than the Property. All approvals, licenses, and permits required from Governmental Authorities under applicable Legal Requirements in connection with the current phase of construction of the Improvements have been obtained and Borrower has no knowledge of any information suggesting that approvals, licenses, and permits for future phases of construction will not be received in a timely manner.
- (g) <u>Services and Utilities</u>. All streets, easements, utilities and related services necessary for the Project and the Property's intended purpose are, or when required will be, available to the Property.
- (h) <u>Enforceability</u>. Each Loan Document executed by Borrower constitutes a legal and binding obligation of, and is valid and enforceable against, Borrower in accordance with the terms thereof (subject to Debtor Relief Laws and general equitable principles) and is not subject to any right of rescission, setoff, counterclaim or defense.
- (i) <u>ERISA</u>. Either (i) neither Borrower nor Guarantor is or will be, or holds or will be treated as holding the assets of, an "employee benefit plan" as defined in Section 3(3) of ERISA, a "governmental plan" as defined in Section 3(32) of ERISA, or a "plan" as defined in Section 4975(e)(1) of the Code; or (ii) the Obligations and the transactions contemplated by the Loan Documents are not and will not be a non-exempt (under a statutory or administrative class or individual exemption) prohibited transaction under ERISA or Section 4975 of the Code, and are not and will not be subject to state statutes applicable to Borrower or Guarantor regulating investments of fiduciaries with respect to governmental plans.
- (j) <u>Employee Benefit Plan Matters</u>. Each Employee Benefit Plan is in material compliance with all applicable requirements under ERISA and the Code. None of the Employee Benefit Plans is subject to the requirements of, or has any material liability under, Section 412 of the Code or Title IV of ERISA or is a "multiemployer plan" as defined in Section 3(37) of ERISA.
- (k) <u>Compliance with Laws and Agreements</u>. Except where the failure to do so, individually or in the aggregate could not reasonably be expected to result in a material adverse effect, Borrower is in compliance with (i) its articles of organization, operating agreement or other organizational documents, (ii) all Legal Requirements applicable to it or its property (including the Property) and (iii) all agreements and other instruments binding upon it or its property (including the Property). No Default or Unmatured Default has occurred and is continuing.
- (I) <u>Anti-Corruption Laws and Sanctions</u>. Each of Borrower, Guarantor, their respective directors and officers, or any Person participating in any capacity in any Transaction is in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects. None of Borrower, Guarantor, their respective directors and officers, or any Person participating in any capacity in any Transaction is a Sanctioned Person.

J.P.Morgan

- (m) <u>Budget</u>. To the best of Borrower's knowledge, the Budget accurately reflects all costs to complete the Improvements and operate the Property through the date currently projected for the commencement of Amazon Tenant's obligations to pay base rent under the Amazon Lease.
- (n) <u>Construction Schedule</u>. To the best of Borrower's knowledge, the Construction Schedule is complete and accurate in all material respects.
- (o) <u>No Prior Conveyance or Limiting Actions</u>. Borrower has not previously conveyed, transferred or assigned the Construction Contract, the Engineer's Contract or the Architect's Contract or any right, title or interest therein.
- (p) <u>Amazon Lease</u>. The Amazon Lease delivered to Lender prior to the Closing Date (i) is a true, correct and complete copy of the Amazon Lease, (ii) is in full force and effect and no defaults or events of default have occurred and are continuing thereunder, and (iii) has not been amended, restated, or otherwise modified, except to the extent approved by Lender in writing.
- 4.02 <u>Nature of Representations and Warranties</u>. All representations and warranties made in this Agreement or any other Loan Document or in any certificate or other document delivered to Lender pursuant to or in connection with this Agreement will be deemed to have been relied upon by Lender notwithstanding any investigation made by Lender or on its behalf.

ARTICLE V: INSURANCE

- 5.01 **Required Insurance Coverage**. Borrower, at its expense, will maintain and provide to Lender copies of policies or other satisfactory evidence of insurance providing the following:
- (a) Commercial General Liability Insurance with limits of not less than \$1,000,000 per occurrence combined single limit and \$2,000,000 in the aggregate for the policy period, or in whatever higher amounts as may be reasonably required by Lender from time to time by notice to Borrower (with deductibles acceptable to Lender in its reasonable judgment), and extended to cover: (i) Contractual Liability assumed by Borrower with defense provided in addition to policy limits for indemnities of the named insured, (ii) if any of the work is subcontracted, Independent Contractors Liability providing coverage in connection with such portion of the work which may be subcontracted, (iii) Broad Form Property Damage Liability, (iv) Products & Completed Operations for coverage, such coverage to apply for two years following completion of construction, (v) waiver of subrogation against all parties named additional insured, (vi) severability of interest provision, and (vii) Personal Injury & Advertisers Liability.
- (b) Automobile Liability including coverage on owned, hired and non-owned automobiles and other vehicles, if used in connection with the performance of the work, with Bodily Injury and Property Damage limits of not less than \$1,000,000.00 per occurrence combined single limit, with a waiver of subrogation against all parties named as additional insured.
- (c) Umbrella/Excess Liability in excess of Commercial General Liability, Automobile Liability and Employers' Liability coverages which is at least as broad as these underlying policies with a limit of liability of \$10,000,000.00.
- (d) All-Risk Property (Special Cause of Loss) Insurance including coverage for loss or damage to the Property by fire and other perils including windstorm, earthquake/earth movement and malicious mischief,

building ordinance extension endorsement (including cost of demolition, increased costs of construction and the value of the undamaged portion of the building and soft costs coverage), and boiler and machinery coverage (if separate policy, that policy must include loss of rents or business interruption coverage), as specified by Lender. All-Risk Policy (Builders' Risk) must include coverage for all off-site Stored Materials and materials in-transit in an amount not less than the dollar limits in Section 2.02(h). The policy must be in an amount not less than the full insurable value on a replacement cost basis of the insured Property and personal property related thereto (without deduction for depreciation). If the policy is a blanket policy covering the Property and one or more other properties, the policy must specify the dollar amount of the total blanket limit of the policy that is allocated to each property including the Property, and the amount so allocated to the Property must not be less than the full insurable value on a replacement cost basis. During any construction period, such policy will be written in the so-called "Builder's Risk Completed Value Non-Reporting Form" with no coinsurance requirement and must contain a provision granting the insured permission to occupy prior to completion. Such policy may not contain an exclusion for terrorist losses.

- (e) Workers' Compensation and Employer's Liability Insurance in accordance with the applicable laws of the state in which the work is to be performed or of the state in which Borrower is obligated to pay compensation to employees engaged in the performance of the work. The policy limit under the Employer's Liability Insurance section must not be less than \$1,000,000.00 for any one accident.
- (f) If at any time, any part of the Improvements, lies within a "special flood hazard area" as designated on maps prepared by the Federal Emergency Management Agency (FEMA), a one hundred year flood plain or other area identified by Lender as having a high or moderate risk of flooding, a flood insurance policy or policies (whether or not coverage is available from the National Flood Insurance Program and whether or not required by the Flood Laws), in form and substance acceptable to Lender covering the Improvements and contents (to the extent the contents secure the Obligations), for the duration of the Loan in an amount at least equal to the full insurable value on a replacement cost basis (without deduction for depreciation) of the Improvements and contents, if applicable, or the amount of the Obligations, whichever is less.
 - (g) Reserved.
- (h) Rent loss or business interruption insurance against loss of income (including, but not limited to, rent, cost reimbursements and all other amounts payable by tenants under Leases or otherwise derived by Borrower from the operation of the Property) arising out of damage to or destruction of the Property by fire or other peril (including earthquake if required) insured against under each policy. The amount of the policy must be in the amount equal to twelve (12) months' projected rentals or gross revenue.
- (i) Such other insurance coverages in such amounts as Lender may reasonably require, which may include errors and omissions insurance with respect to contractors, architects and engineers.
- Policy Requirements. All insurance policies must (a) be issued by an insurance company licensed to do business in the state where the Property is located having a rating of "A-" VIII or better by A.M. Best Co., in Best's Rating Guide, (b) name "JPMorgan Chase Bank, N.A. and any and all subsidiaries and their successors and/or assigns as their interests may appear" as additional insured on all liability insurance and as mortgagee and loss payee on all All-Risk Property, flood insurance, earthquake insurance and rent loss or business interruption insurance (whether or not required hereunder), (c) be endorsed to show that Borrower's insurance will be primary and all insurance carried by Lender is strictly excess and secondary and will not contribute with

Borrower's insurance, (d) provide that Lender is to receive at least thirty (30) days (ten (10) days in the event of failure to pay premium) written notice prior to non-renewal or cancellation, (e) be evidenced by a certificate of insurance to be provided to Lender along with a copy of the policy for All-Risk Property coverage or such other evidence of insurance acceptable to Lender in its reasonable discretion, (f) include either policy or binder numbers on the ACORD form, and (g) be in form and amounts acceptable to Lender in its reasonable discretion; provided, however, that with respect to any flood insurance required hereunder, acceptable proof of coverage will consist of a copy of the insurance policy, the declarations page of the insurance policy or an application plus proof of premium payment (with a copy of the policy or declarations page provided to Lender within 30 days thereafter) and may not include ACORD or other forms of certificates of insurance. Lender, at its option and upon notice to Borrower, may retain at Borrower's expense, an insurance consultant to review the insurance for the Property to confirm that it complies with the terms and conditions herein.

- 5.03 Evidence of Insurance; Payment of Premiums. Borrower will deliver to Lender, at least five (5) days before the expiration of an existing policy, evidence acceptable to Lender of the continuation of the coverage of the expiring policy. Nothing contained in this Section will require Lender to incur any expense or take any action hereunder, and inaction by Lender will never be considered a waiver of any right accruing to Lender on account on this Section. Borrower agrees to pay all premiums on such insurance as they become due, and will not permit any condition to exist on or with respect to the Project or the Property which would wholly or partially invalidate any insurance thereon.
- Collateral Protection. Unless Borrower provides Lender with evidence satisfactory to Lender of the insurance coverage required by this Agreement, Lender may purchase insurance at Borrower's expense to protect Lender's interest in the Property. This insurance may, but need not, protect Borrower's interest in the Property. The coverages that Lender purchases may not pay any claim that Borrower makes or any claim that is made against Borrower in connection with the Property. Borrower or Lender (as appropriate) may later cancel any insurance purchased by Lender, but only after Lender receives satisfactory evidence that Borrower has obtained insurance as required by this Agreement. If Lender purchases insurance for the Property, Borrower will be responsible for the costs of that insurance, including any charges imposed by Lender in connection with the placement of insurance, until the effective date of the cancellation or expiration of such insurance. Any amounts so disbursed by Lender pursuant to this Section will be repaid by Borrower within 10 days after written demand. The costs of the insurance may, at Lender's discretion, be added to Borrower's total principal obligation owing to Lender, and in any event will be secured by the liens on the Property created by the Loan Documents. It is understood and agreed that (i) the costs of insurance obtained by Lender may be more than the costs of insurance Borrower may be able to obtain on its own, and (ii) in the case of flood insurance, the amount of coverage may be more than required by the Flood Laws.
- No Liability; Assignment. Lender will not by the fact of approving, disapproving, accepting, preventing, obtaining or failing to obtain any such insurance, incur any liability for the form or legal sufficiency of insurance contracts, solvency of insurers, or payment of losses, and Borrower hereby expressly assumes full responsibility therefor and all liability, if any, thereunder. Borrower hereby absolutely assigns and transfers to Lender all of Borrower's right, title and interest in and to any unearned premiums paid on policies and any claims thereunder and Lender will have the right, but not the obligation, to assign any then existing claims under the same to any purchaser of the Property at any foreclosure sale; provided, however, that so long as no Default exists and is continuing hereunder, Borrower will have the right under a license granted hereby, and Lender hereby grants to Borrower a license, to exercise rights under said policies and in and to said premiums subject to the provisions of this Agreement. In the event of a foreclosure of the Security Instrument, or other

transfer of title to the Property in extinguishment in whole or in part of the Loan, all right, title and interest of Borrower in and to the insurance policies then in force and all proceeds payable thereunder will thereupon vest in the purchaser at such foreclosure or Lender or other transferee in the event of such other transfer of title.

5.06 **No Separate Insurance**. Borrower shall not carry any separate insurance on the Property concurrent in kind or form with any insurance required hereunder or contributing in the event of loss without Lender's prior written consent, and any such policy must have attached a standard non-contributing mortgagee clause, with loss payable to Lender, and must otherwise meet all other requirements herein.

ARTICLE VI: DEFAULTS AND REMEDIES

- 6.01 **Defaults**. Any of the following events, after passage of the applicable cure period set forth below, will constitute a "Default" hereunder:
- (a) <u>Failure to Make Payment</u>. The failure by Borrower to pay in full all outstanding principal, all accrued and unpaid interest and all other amounts owing under the Loan Documents on the Maturity Date; or the failure by Borrower to pay in full any installment of principal of the Loan, any interest on the Loan or any fees or any other amounts due under the Loan Documents (other than payments due on the Maturity Date) when due and such failure continues unremedied for a period of five (5) days after the due date; or the failure by Borrower to make any other payment or deposit required hereunder or under any of the other Loan Documents within the period in the Loan Documents, or if no period is in the Loan Documents, then within five (5) Business Days after demand;
- (b) Involuntary Bankruptcy and Other Legal Proceedings. An involuntary proceeding is commenced or an involuntary petition is filed seeking (i) liquidation, reorganization or other relief in respect of Borrower or Guarantor or its debts, or of a substantial part of its assets, under any Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect or (ii) the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for Borrower or Guarantor or for a substantial part of its assets, and, in any such case, such proceeding or petition continues undismissed for sixty (60) days or an order or decree approving or ordering any of the foregoing is entered;
- (c) Voluntary Bankruptcy and Other Legal Proceedings. Borrower or Guarantor (i) voluntarily commences any proceeding or files any petition seeking liquidation, reorganization or other relief under any Federal, state or foreign bankruptcy, insolvency, receivership or similar law now or hereafter in effect, (ii) consents to the institution of, or fails to contest in a timely and appropriate manner, any proceeding or petition described in Section 6,01(b), (iii) applies for or consents to the appointment of a receiver, trustee, custodian, sequestrator, conservator or similar official for Borrower or Guarantor or for a substantial part of its assets, (iv) files an answer admitting the material allegations of a petition filed against it in any such proceeding, (v) makes a general assignment for the benefit of creditors, or (vi) takes any action for the purpose of effecting any of the foregoing;
- (d) <u>Unable to Pay Debts</u>. The admission in writing by Borrower or Guarantor that it is unable to pay its debts as they mature or that it is generally not paying its debts as they mature;
- (e) <u>Preservation of Existence; Liquidation of Borrower or Guarantor</u>. The liquidation, termination, windingup or dissolution of Borrower or Guarantor or any other failure by Borrower or Guarantor to preserve its existence as an entity duly organized, validly existing and in good standing (if applicable) under the laws of jurisdiction of its organization or formation;

- (f) <u>Transfer or Encumbrance of Interest in the Property</u>. The sale, lease (except as permitted under this Agreement), exchange, conveyance, transfer, mortgage, assignment, pledge or encumbrance, either voluntarily or involuntarily, or the agreement to do so, of any right, title or interest of Borrower in and to the Property or any portion of the Property, which occurrence is not rendered ineffective within ten (10) days; provided, however, that Borrower is permitted to replace defective, obsolete or worn out personal property, and Borrower is permitted to grant and/or record Permitted Encumbrances;
- (g) <u>Transfer or Encumbrance of Interest in Borrower</u>. The sale, exchange, conveyance, transfer, mortgage, assignment, pledge or encumbrance, either voluntarily or involuntarily, or the agreement to do so, of any direct or indirect ownership interest in Borrower; or any change (whether voluntary or involuntary) in the management or control of Borrower; or any division of Borrower (whether pursuant to a "plan of division" or other similar arrangement); provided, however, that a Permitted Transfer will not constitute a Default under this Section 6.01(g);
- (h) <u>Levy; Attachment; Seizure</u>. The levy, attachment or seizure pursuant to court order of (i) any right, title or interest of Borrower in and to the Property or any portion of the Property or (ii) any direct or indirect ownership interests in Borrower, if such order is not vacated and the proceeding in which it was entered is not dismissed within thirty (30) days of the entry of such order;
- (i) <u>Failure of Representations</u>. Any representation or warranty contained herein or in any of the other Loan Documents, or in any certificate or other document executed by Borrower or Guarantor and delivered to Lender pursuant to, or in connection with, this Agreement, is not true and correct in all material respects, or omits to state a material fact necessary to make such representation not misleading, in each case, as of the date made or deemed made;
- (j) Claims; Liens; Encumbrances; Stop Notices. Unless Borrower is contesting the same in accordance with the provisions of Section 3.01(c), the filing of any claim of Lien or encumbrance against the Property or any part of the Property that is not released of record within fifteen (15) days after the earlier of Borrower's actual knowledge of such filing of or notice from Lender to Borrower; or the service on Lender of any stop payment notice or notice to disburser of funds to withhold funds, which is not nullified within fifteen (15) days after the date of such service;
- (k) Permits; Utilities; Insurance. (i) The failure of Borrower to keep in full force and effect any material permit, license, consent or approval required for the Project (including, without limitation, any temporary or final certificate of occupancy) that is not fully reinstated within thirty (30) days after the earlier of the date Lender gives Borrower notice of the lapse of effectiveness of such material permit, license, consent or approval or the date Borrower obtains actual knowledge of such lapse of effectiveness; or (ii) the curtailment in availability to the Property of utilities or other public services necessary for the full occupancy and utilization of the Property that is not restored to full availability within thirty (30) days after the earlier of the date Lender gives Borrower notice of such curtailment of availability or the date Borrower obtains actual knowledge of such curtailment of availability unless the same is due to an areawide shutdown by or effecting the applicable utility provider and not caused by an action or inaction of Borrower; or (iii) the failure by Borrower to maintain any insurance required under Section 5.01;
- (I) <u>Effectiveness of Loan Documents; Guaranty Revocation</u>. The cessation, for any reason, of any Loan Document to be in full force and effect in all material respects; the failure of any Lien intended to be created by

the Loan Documents to exist or to be valid and perfected; or the revocation by Guarantor of the Completion Guaranty, the Limited Guaranty, or any other Loan Document executed by Guarantor;

- (m) ERISA. Any breach of the provisions of Sections 3.09, 4.01(i) or 4.01(j);
- (n) <u>Prohibited Distributions</u>. Any breach of the provisions of Section 3.06 occurs; which is not cured by Borrower within five (5) days after such breach;
- (o) <u>Special Purpose Entity Requirements</u>. Any breach of the provisions of Section 3.03 occurs which is not cured by Borrower within five (5) days after such breach;
- (p) <u>Judgments</u>. Any judgment or order for the payment of money in excess of \$250,000 is rendered against Borrower or in excess of \$10,000,000 against Guarantor and either (i) enforcement proceedings have been commenced by a creditor upon such judgment, or (ii) there is a period of fifteen (15) days during which a stay of enforcement of such judgment or order, by reason of a pending appeal or otherwise, is not in effect;
- (q) <u>Amazon Lease</u>. (i) The occurrence of a default beyond applicable cure periods by Borrower under the Amazon Lease or (ii) the termination of the Amazon Lease without Lender's prior written consent.
- (r) <u>Guarantor Cross-Default</u>. Failure by Guarantor to pay within any applicable notice and grace period after the same becomes due any indebtedness, whether direct or contingent, owing to Lender or any Affiliate of Lender (other than the Loan) or any indebtedness, whether direct or contingent, in an outstanding principal amount of \$25,000,000 or more in the aggregate owing to any other Person or Persons;
- (s) <u>Borrower Cross-Default</u>. Failure by Borrower to pay within any applicable notice and grace period after the same becomes due (whether on a scheduled payment date or earlier date of acceleration) any indebtedness, whether direct or contingent, owing to Lender or any Affiliate of Lender (other than the Loan) or to any other Person or Persons;
 - (t) Anti-Corruption Laws and Sanctions. Any breach of the covenants in Section 4.01(I);
- (u) Other Loan Document. Any default occurs under any other Loan Document and the same continues beyond any applicable notice and grace period;
- (v) <u>Cessation of Construction</u>. Cessation of the work of construction prior to the completion of the Improvements for a continuous period of thirty (30) days or more except as a result of Force Majeure Causes, or the obtaining by any Person of any order or decree in any court of competent jurisdiction enjoining the construction of the Improvements which order or decree is not vacated within thirty (30) days after the granting of such order or decree:
- (w) <u>Change in Contractor</u>. The occurrence of any change in the Contractor without Lender's prior written consent, which consent shall not be unreasonably withheld, conditioned or delated;
- (x) <u>Failure to Satisfy Construction Obligations</u>. Borrower's failure to cause the Improvements to be Substantially Completed as and when required by Section 3.12(b); or
- (y) <u>Failure to Perform Covenants</u>. The failure of Borrower to fully perform any and all covenants and agreements under this Agreement or under any of the other Loan Documents, and, with respect to covenants and agreements other than those specifically referenced in this Section 6.01, or for which another cure period is

provided, such failure ("other default") is not cured by Borrower within thirty (30) days after Lender gives notice to Borrower, unless (i) such other default, by its nature, is not capable of being cured within such period, and (ii) within such period, Borrower commences to cure such other default and thereafter diligently prosecutes the cure, and (iii) Borrower causes such other default to be cured no later than ninety (90) days after the date of such notice from Lender.

- Acceleration. If any Default described in Section 6.01(b) or (c) occurs with respect to Borrower, the obligation of Lender to lend or disburse any principal of the Loan will automatically terminate and the Obligations will immediately become due and payable without any election or action on the part of Lender. During the continuance of any other Default, Lender may terminate or suspend the obligation of Lender to lend or disburse any principal of the Loan, or may declare the Obligations to be due and payable, or both, whereupon the Obligations shall become immediately due and payable, without presentment, demand, protest or notice of any kind, all of which Borrower hereby expressly waives.
- Additional Lender Rights. During the continuance of a Default, without waiving the Default or any right of acceleration or foreclosure under the Loan Documents which Lender may have by reason of such Default or any other right Lender may have against Borrower because of said Default, Lender will have the right (but not the obligation) to take such actions and make such payments as shall be necessary to cure such Default, including the making of any Advance. Borrower will reimburse Lender upon demand for all amounts so expended, and such obligation will constitute an Obligation of Borrower.
- 6.04 **Application of Insufficient Funds**. Lender will have the right to apply repayments and proceeds of collateral to the Obligations in any order, in its sole discretion.
- 6.05 Right of Setoff. If a Default has occurred and is continuing, Lender and its Affiliates is hereby authorized at any time, to the fullest extent permitted by law, to setoff and apply any and all deposits (general or special, time or demand, provisional or final) at any time held and other obligations at any time owing by Lender or its Affiliate to or for the credit or the account of Borrower against any or all of the Obligations, irrespective of whether or not Lender has made any demand under this Agreement or any other Loan Document and although such obligations of Borrower may be unmatured. The rights of Lender under this Section are in addition to other rights and remedies (including other rights of setoff) which Lender may have.
- 6.06 Right to Complete Construction. Without limitation of any other right or remedy available to Lender under the Loan Documents or applicable law, upon the occurrence and during the continuance of a Default, Lender shall have the right in person or by agent to take possession of the Property and perform any and all work and labor necessary to complete the Improvements and protect the Property from injury. All reasonable sums so expended by Lender shall be deemed to have been paid to Borrower and constitute Obligations. Effective upon the occurrence and during the continuance of a Default, Borrower hereby constitutes and appoints Lender its true and lawful attorney-in-fact, with full power of substitution, to complete the Improvements in the name of Borrower. Borrower hereby empowers said attorney to: (a) use any funds of Borrower, including any funds which may remain undisbursed under this Agreement for the purpose of completing the Improvements; (b) make such additions, changes and corrections in the Plans and Specifications as Lender reasonably deems appropriate; (c) employ such contractors, subcontractors, agents, architects and inspectors as shall be required for said purposes; (d) pay, settle or compromise all existing bills and claims which may be Liens against the Property, or as may be necessary or desirable for such completion of the Improvements or for clearance of title; (e) execute all applications and certificates in the name of Borrower which may be required by any of the contract documents; (f) prosecute and defend all actions or

proceedings in connection with the Property or the construction of the Improvements and take such action and require such performance as it deems necessary under any bond or guaranty of completion; and (g) do any and every act which Borrower might do on its own behalf. It is further understood and agreed that this power of attorney, which shall be deemed to be a power coupled with an interest, cannot be revoked.

ARTICLE VII: MISCELLANEOUS

7.01 **Notices**.

- (a) <u>Generally</u>. Except in the case of notices and other communications expressly permitted or required to be given by electronic communication (and subject to paragraph (b) below), all notices and other communications provided for herein will be in writing and will be delivered by hand or overnight courier service, or mailed by certified or registered mail, as follows:
- (i) if to Borrower, to Riverbend Old Statesville, LLC c/o INDUS Realty, LLC, 204 West Newberry Road, Bloomfield, Connecticut 06002 Attention: Anthony J. Galici, Chief Financial Officer (Email: agalici@indusrt.com with copies to (i) the Borrower at the foregoing address Attention: Thomas M. Daniells, Esquire, General Counsel (Email: tdaniells@indusrt.com, and (ii) Murtha Cullina LLP, 280 Trumbull Street, Hartford, Connecticut 06103 Attention: Frank J. Saccomandi, III, Esquire (Email: fsaccomandi@murthalaw.com); and;
- (ii) if to Lender: (A) for financial reporting required under Section 3.08, at REB.Compliance@jpmorgan.com; and (B) for all other notices and communications, at 201 N. Central Avenue, 20th Floor, Phoenix, AZ 85004, Attention of Ryan Dempsey (Email Address ryan.m.dempsey@jpmorgan.com.

Notices sent by hand or overnight courier service, or mailed by certified or registered mail, will be deemed to have been given when received.

- (b) <u>Electronic Notices</u>. Lender or Borrower may, in its discretion, agree to accept notices and other communications by using electronic communications pursuant to procedures approved by them; provided that approval of such procedures may be limited to particular notices or communications. Unless Lender otherwise prescribes, (i) notices and other communications sent to an e-mail address will be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), and (ii) notices or communications posted to an Internet or intranet website will be deemed received upon the "receipt" by the intended recipient, at its e-mail address as described in clause (i), of notification that such notice or communication is available and identifying the website address, provided that, for both clauses (i) and (ii) above, if such notice, email or other communication is not sent during the normal business hours of the recipient, such notice or communication will be deemed to have been sent at the opening of business on the next Business Day.
- (c) <u>Changes in Address</u>. Any party may change its address or email address for notices and other communications hereunder by notice to the other parties hereto.

7.02 Waivers; Amendments.

(a) <u>No Deemed Waivers; Remedies Cumulative</u>. No failure or delay by Lender in exercising any right or power under this Agreement will operate as a waiver, nor will any single or partial exercise of any such right or power, or any abandonment or discontinuance of steps to enforce such a right or power, preclude any other or further exercise thereof or the exercise of any other right or power. The rights and remedies of Lender under this

Agreement are cumulative and are not exclusive of any rights or remedies that they would otherwise have. No waiver of any provision of this Agreement or consent to any departure by Borrower therefrom will in any event be effective unless the same will be permitted by Section 7.02(b), and then such waiver or consent will be effective only in the specific instance and for the purpose for which given. Without limiting the generality of the foregoing, the making of any Advance will not be construed as a waiver of any Default or Unmatured Default, regardless of whether Lender may have had notice or knowledge of such Default or Unmatured Default at the time.

(b) <u>Waivers and Amendments</u>. No provision of this Agreement or any other Loan Document may be waived, amended or modified except pursuant to an agreement or agreements in writing entered into by Borrower and Lender.

7.03 Expenses; Limitation of Liability; Indemnity.

- (a) Expenses. Borrower will pay (i) all reasonable out -of -pocket expenses incurred by Lender and its Affiliates, including Appraisal fees, inspection fees, Inspecting Professional charges, title and escrow charges, cost of any third party insurance consultant report and the reasonable fees, charges and disbursements of counsel for Lender, in connection with the preparation and administration of this Agreement and the other Loan Documents, any approval or consent requested by Borrower or Guarantor, or any extensions, amendments, modifications or waivers of such provisions, and (ii) all out-of-pocket expenses incurred by Lender, including the reasonable fees, charges and disbursements of any counsel for Lender, in connection with the enforcement or protection of its rights in connection with this Agreement and the other Loan Documents, including its rights under this Section, or in connection with the Loan, including all such out-of-pocket expenses incurred during any workout, restructuring or negotiations in respect of the Loan.
- (b) <u>Limitation of Liability</u>. To the extent permitted by applicable law (i) Borrower shall not assert, and Borrower hereby waives any claim against Lender and any Related Party of Lender ("<u>Lender-Related Person</u>") for any Liabilities arising from the use by others of information or other materials (including any personal data) obtained through telecommunications, electronic or other information transmission systems (including the Internet), and (ii) no party to this Agreement shall assert, and each such party waives, any Liabilities against any other party to this Agreement, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document (other than the Environmental Indemnity Agreement) or any related agreement or document, the Transactions, the Loan or the use of the proceeds of the Loan; provided that, nothing in this Section 7.03(b) shall relieve Borrower of any obligation it may have to indemnify an Indemnitee, as provided in Section 7.03(c), against any special, indirect, consequential or punitive damages asserted against such Indemnitee by a third party.
- (c) <u>Indemnity</u>. Borrower will indemnify Lender and each Related Party of Lender (each an "<u>Indemnitee</u>") against, and hold each Indemnitee harmless from, any and all losses, claims, damages, penalties, taxes, judgments, liabilities and related expenses, including the reasonable fees, charges and disbursements of any counsel for any Indemnitee and all expenses of litigation or preparation therefor (whether or not the Indemnitee is a party) (collectively, "<u>Liabilities</u>"), incurred by, or asserted against, any Indemnitee, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any related agreement or document, the performance by the parties of their respective obligations under such agreements and documents or the inaccuracy of any of Borrower's representations, warranties or covenants therein, (ii) the Loan or the use

of the proceeds of the Loan, (iii) any actual or prospective claim, litigation, investigation, defense, settlement or proceeding relating to any of the foregoing (including, if applicable, in connection with obtaining any prohibited transaction exemption under ERISA or the Code necessary in Lender's judgment by reason of the inaccuracy of the representations and warranties in Sections 4.01(i) and (j) or a breach of the provisions set forth in Section 3.09); provided that such indemnity will not, as to any Indemnitee, be available to the extent that such Liabilities or related expenses are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted primarily from the gross negligence or willful misconduct of an Indemnitee. The indemnity in this Section 7.03(b) shall not apply to any Liabilities, which are the subject of the Environmental Indemnity Agreement, it being the intention of the parties hereto that Borrower's liability for environmental matters be governed exclusively by the Environmental Indemnity Agreement and not by this Agreement. The obligations of Borrower under this Section 7.03(c) shall survive the termination of this Agreement and the payment in full of the Obligations.

(d) Payment. All amounts due under this Section 7.03 will be payable promptly after written demand.

7.04 <u>Successors and Assigns</u>.

(a) <u>Binding Effect</u>. The provisions of this Agreement are binding upon, and inure to the benefit of, the parties and their respective successors and assigns permitted under this Agreement. Except as permitted under this Agreement, Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of Lender, in Lender's sole discretion (and any attempted assignment or transfer by Borrower without such consent shall be void). Nothing in this Agreement, expressed or implied, will be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, participants of Lender and, to the extent expressly contemplated hereby, the Related Parties of Lender) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) Assignment by Lender.

- (i) Subject to the conditions in Section 7.04(b)(ii), Lender may assign to one or more Persons all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loan at the time owing to it) with the prior written consent (such consent not to be unreasonably withheld) of Borrower; provided no consent of Borrower will be required for an assignment to an Affiliate of Lender or an Approved Fund or, if a Default has occurred and is continuing, any other assignee.
- (ii) Subject to Lender's notification to Borrower of an assignment, assignee will be a party to this Agreement and, to the extent of the interest assigned, have the rights and obligations of a Lender under this Agreement, and Lender will, to the extent of the interest assigned, be released from its obligations under this Agreement (and, in the case of an assignment covering all of Lender's rights and obligations under this Agreement, such Lender will cease to be a party but shall continue to be entitled to the benefits of Section 7.03). Borrower agrees to execute any amendment and/or any other document that may be necessary to effectuate such an assignment, including an amendment to this Agreement to provide for multiple lenders and an administrative agent to act on behalf of such lenders and other Loan Syndications and Trading Association market standard provisions acceptable to Borrower and Lender.
- (c) <u>Pledges by Lender</u>. Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank, and this Section will not apply to any such pledge or assignment of a

security interest; provided that no such pledge or assignment of a security interest will release Lender from any of its obligations under this Agreement or substitute any such pledgee or assignee for Lender as a party.

Survival. All covenants, agreements, representations and warranties made by Borrower herein and in the certificates or other instruments delivered in connection with or pursuant to this Agreement and the other Loan Documents will be considered to have been relied upon by Lender and will survive the execution and delivery of this Agreement and the other Loan Documents and the making of the Loan, regardless of any investigation made by Lender or on its behalf and notwithstanding that Lender may have had notice or knowledge of any Default or Unmatured Default or incorrect representation or warranty at the time any credit is extended hereunder, and will continue in full force and effect as long as the principal of or any accrued interest on any portion of the Loan or any fee or any other amount payable under this Agreement is outstanding. The provisions of Section 7.03 shall survive and remain in full force and effect regardless of the consummation of the Loan, the repayment of the Loan, the expiration or termination of the Commitment or the termination of this Agreement or any provision of this Agreement.

7.06 Counterparts; Integration; Effectiveness; Electronic Execution.

- (a) This Agreement may be executed in counterparts (and by different parties hereto on different counterparts), each of which will constitute an original, but all of which when taken together will constitute a single contract. This Agreement and the other Loan Documents constitute the entire contract among the parties relating to the subject matter of this Agreement and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter of this Agreement.
- (b) Delivery of an executed counterpart of a signature page of (x) this Agreement, (y) any other Loan Document and/or (z) any document, amendment, approval, consent, information, notice (including, for the avoidance of doubt, any notice delivered pursuant to Section 7.01), certificate, request, statement, disclosure or authorization related to this Agreement, any other Loan Document and/or the transactions contemplated hereby and/or thereby (each an "Ancillary Document") that is an Electronic Signature transmitted by telecopy, emailed pdf. or any other electronic means that reproduces an image of the actual executed signature page shall be effective as delivery of a manually executed counterpart of this Agreement, such other Loan Document or Ancillary Document, as applicable. The words "execution," "signed," "signature," "delivery," and words of like import in or relating to any document to be signed in connection with this Agreement, any other Loan Document and/or such Ancillary Document shall be deemed to include Electronic Signatures, deliveries or the keeping of records in any electronic form (including deliveries by telecopy, emailed pdf, or any other electronic means that reproduces an image of an actual executed signature page), each of which shall be of the same legal effect, validity or enforceability as a manually executed signature, physical delivery thereof or the use of a paper-based recordkeeping system, as the case may be; provided that nothing herein shall require Lender to accept Electronic Signatures in any form or format without its prior written consent and pursuant to procedures approved by it; provided, further, without limiting the foregoing, (i) to the extent Lender has agreed to accept any Electronic Signature, Lender shall be entitled to rely on such Electronic Signature purportedly given by or on behalf of Borrower or Guarantor without further verification thereof and without any obligation to review the appearance or form of any Electronic Signature and (ii) upon the request of Lender, any Electronic Signature shall be promptly followed by a manually executed counterpart. Without limiting the generality of the foregoing, Borrower hereby (i) agrees that, for all purposes, including without limitation, in connection with any workout, restructuring, enforcement of remedies, bankruptcy proceedings or litigation among Lender, Borrower and Guarantor, Electronic Signatures transmitted by telecopy, emailed pdf. or any other electronic means that

reproduces an image of an actual executed signature page and/or any electronic images of this Agreement, any other Loan Document and/or Ancillary Document shall have the same legal effect, validity and enforceability as any paper original, (ii) Lender, may, at its option, create one or more copies of this Agreement, any other Loan Document and/or Ancillary Document in the form of an imaged electronic record in any format, which shall be deemed created in the ordinary course of such Person's business, and destroy the original paper document (and all such electronic records shall be considered an original for all purposes and shall have the same legal effect, validity and enforceability as a paper record), and (iii) waives any argument, defense or right to contest the legal effect, validity or enforceability of the Agreement, such other Loan Documents and Ancillary Document based solely on the lack of paper original copies of this Agreement, any other Loan Document and/or Ancillary Document, respectively, including with respect to any signature pages thereto, and (iv) waives any claim against any Lender-Related Person for any Liabilities arising solely from Lender's reliance on or use of Electronic Signatures and/or transmissions by telecopy, emailed pdf. or any other electronic means that reproduces an image of an actual executed signature page, including any Liabilities arising as a result of the failure of Borrower to use any available security measures in connection with the execution, delivery or transmission of any Electronic Signature.

7.07 **Severability**. Any provision of this Agreement or any other Loan Document held to be invalid, illegal or unenforceable in any jurisdiction will, as to such jurisdiction, be ineffective to the extent of such invalidity, illegality or unenforceability without affecting the validity, legality and enforceability of the remaining provisions hereof or thereof; and the invalidity of a particular provision in a particular jurisdiction shall not invalidate such provision in any other jurisdiction.

7.08 Governing Law; Jurisdiction; Consent to Service of Process.

- (a) <u>Governing Law</u>. This Agreement will be construed in accordance with and governed by the internal laws (and not the law of conflicts) of the State of New York, except preempted by federal laws applicable to national banks.
- (b) Consent to Jurisdiction. Each of the parties to this Agreement hereby (i) submits, for itself and its property, to the nonexclusive jurisdiction of any United States Federal or State court sitting in New York, New York or Mecklenburg County, North Carolina, and any appellate court in such jurisdiction, in any action or proceeding arising out of or relating to any Loan Documents, or for recognition or enforcement of any judgment, (ii) agrees that all claims in respect of any such action or proceeding may (and any such claims, cross-claims or third party claims brought against Lender or any of its Related Parties may only) be heard and determined in such State or, to the extent permitted by law, in such Federal court, and (iii) agrees that a final judgment in any such action or proceeding will be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law. Nothing in this Agreement or any other Loan Document will affect any right that Lender may otherwise have to bring any action or proceeding relating to this Agreement or any other Loan Document against Borrower or Guarantor or its properties in the courts of any jurisdiction.
- (c) <u>Waiver of Objection to Venue</u>. Each of the parties to this Agreement waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the venue of any suit, action or proceeding arising out of or relating to this Agreement or any other Loan Document in any court referred to in Section 7.08(b). Each of the parties hereto hereby waives, to the fullest extent permitted by law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such court.

- (d) <u>Service of Process</u>. Each party to this Agreement consents to service of process in the manner provided for notices in Section 7.01. Nothing in this Agreement will affect the right of any party to this Agreement to serve process in any other manner permitted by law.
- 7.09 WAIVER OF JURY TRIAL. EACH PARTY HERETO HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY LOAN DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.
- **Confidentiality.** Lender agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its and its Affiliates' directors, officers, employees and agents, including accountants, legal counsel and other advisors (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any Governmental Authority (including any selfregulatory authority, such as the National Association of Insurance Commissioners), (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) to any other party to this Agreement, (e) in connection with the exercise of any remedies hereunder or under any other Loan Document or any suit, action or proceeding relating to the Obligations or the enforcement of rights under the Loan Documents, (f) subject to an agreement containing provisions substantially the same as those of this Section 7.10, to any assignee of or participant in, or any prospective assignee of or participant in, any of its rights or obligations under this Agreement, (g) with the consent of Borrower, (h) to holders of equity interests in Borrower, or (i) to the extent such Information (x) becomes publicly available other than as a result of a breach of this Section or (y) becomes available to Lender on a non-confidential basis from a source other than Borrower. For the purposes of this Section, "Information" means all information received from Borrower relating to Borrower or its business, other than any such information that is available to Lender on a non-confidential basis prior to disclosure by Borrower and other than information pertaining to this Agreement routinely provided by arrangers to data service providers, including league table providers, that serve the lending industry: provided that, in the case of information received from Borrower after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section will be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information. Lender acknowledges that information (as defined in this Section) furnished to it pursuant to this Agreement may include material non-public information concerning Borrower, Guarantor, and their Related Parties or their respective securities, and confirms that it has developed compliance procedures regarding the use of material non-public information and that it will handle such material non-public information in accordance with those procedures and applicable law, including federal and state securities laws.
- 7.11 <u>Interest Rate Limitation</u>. If any amount contracted for, charged or received with respect to the Loan or any prepayment by Borrower results in the computation or earning of any interest in excess of that permitted by

law, then (i) such excess sum received by Lender will be credited as a payment of principal unless Borrower notifies Lender in writing that Borrower elects to have such excess returned to Borrower, and (ii) the provisions of this Agreement and the other Loan Documents will immediately be deemed reformed to comply with applicable law and to permit the recovery of the fullest amount otherwise provided by this Agreement and the other Loan Documents. All amounts treated as interest on the Loan under applicable law will be spread throughout the full term of the Loan until paid in full, so that the rate or amount of interest applicable to the Loan does not exceed the usury ceiling from time to time in effect.

- Replacement Documentation. Promptly upon request by Lender, Borrower will, at Borrower's sole expense, (i) promptly correct any material defect or error that may be discovered in any Loan Document or in the execution, acknowledgment, filing or recordation thereof, and (ii) do, execute, acknowledge, deliver, record, re-record, file, re-file, register and re-register any and all such further acts, deeds, certificates, assurances and other instruments as Lender may reasonably require from time to time in order to (A) carry out more effectively the purposes of the Loan Documents, (B) perfect and maintain the validity, effectiveness and priority of any of the Loan Documents and any of the Liens intended to be created thereunder, (C) provide such certificates, documents, reports, information, affidavits and other instruments and do such further acts deemed necessary, desirable or proper by Lender to comply with the requirements of any Governmental Authority having jurisdiction over Lender. Upon any failure by Borrower to do so, Lender may make, execute and record any and all such instruments, certificates and other documents for and in the name of Borrower, all at the sole expense of Borrower, and Borrower hereby appoints Lender the agent and attorney-in-fact of Borrower to do so, this appointment being coupled with an interest and being irrevocable.
- 7.13 **Statements**. Lender may from time to time provide Borrower with account statements or invoices with respect to any of the Obligations (the "Statements"). Lender is under no duty or obligation to provide Statements, which if provided, will be solely for Borrower's convenience. Statements may contain estimates of the amounts owed during the relevant billing period, whether of principal, interest, fees or other Obligations. If Borrower pays the full amount indicated on a Statement on or before the due date indicated on such Statement, Borrower will not be in default of payment with respect to the billing period indicated on such Statement; provided, that acceptance by Lender of any payment that is less than the total amount actually due at that time (including, but not limited to, any past due amounts) will not constitute a waiver of Lender's right to receive payment in full as provided by the terms of this Agreement and the other Loan Documents.
- 7.14 LIBOR Notification. The interest rate on the Loan is determined by reference to the LIBOR Rate, which is derived from the London interbank offered rate ("LIBOR"). LIBOR is intended to represent the rate at which contributing banks may obtain short-term borrowings from each other in the London interbank market. LIBOR is currently the subject of regulatory reform and regulators have signaled the need to use alternative benchmark reference rates for LIBOR. On March 5, 2021, the U.K. Financial Conduct Authority ("FCA") publicly announced that (i) immediately after December 31, 2021, publication of the 1-week and 2-month U.S. Dollar LIBOR settings will permanently cease; (ii) immediately after June 30, 2023, publication of the overnight and 12-month U.S. Dollar LIBOR settings will permanently cease; and (iii) immediately after June 30, 2023, the 1-month, 3-month and 6-month U.S. Dollar LIBOR settings will cease to be provided or, subject to the FCA's consideration of the case, be provided on a synthetic basis and no longer be representative of the underlying market and economic reality they are intended to measure and that representativeness will not be restored. There is no assurance that dates announced by the FCA will not change or that the administrator of LIBOR and/or regulators will not take further action that could impact the availability, composition, or characteristics of LIBOR or the currencies and/or tenors for which LIBOR is published, and Borrower and Guarantor should

consult their own advisors to stay informed of any such developments. Public and private sector industry initiatives are currently underway to implement new or alternative reference rates to be used in place of LIBOR. In the event LIBOR is no longer available (or in certain other circumstances), Section 2.06 of this Agreement provides a mechanism for determining an alternative rate of interest. Lender will inform Borrower of any change to the reference rate upon which the interest rate of the Loan is based. However, Lender does not warrant or accept any responsibility for, and shall not have any liability with respect to, the administration, submission, performance or any other matter related to LIBOR or any other rates in the definition of LIBOR Rate or with respect to any alternative, successor or replacement reference rates, including whether the composition or characteristics of any such alternative, successor or replacement reference rates will be similar to or produce the same value or economic equivalence as LIBOR and/or the LIBOR Rate or have the same volume or liquidity as did LIBOR prior to its discontinuance or unavailability.

- Release of Roadway. Lender acknowledges that in accordance with the plans for the Improvements which have been approved by all applicable Governmental Authorities, Borrower is obligated to construct, and upon completion thereof, dedicate to the applicable Governmental Authority, a roadway along the southern boundary of the Property (as shown on the Plans and Specifications) which is to be a public road. Lender agrees that, upon request of Borrower once such roadway has been completed in accordance with the approved Plans and Specifications, Lender, at no expense to Lender, shall cooperate with Borrower in its dedication of such roadway to the applicable Governmental Authority and shall execute and deliver, and cause the Trustee to execute and deliver, any release or other documentation which is reasonably required to release such roadway from the lien of the Security Instrument; provided, however, that Lender's obligation to release such roadway from the lien of the Security Instrument is expressly conditioned upon the satisfaction of the following conditions precedent:
- (a) Lender shall have received evidence satisfactory to Lender that: (i) the portion of the Property to be released and the portion of the Property which shall remain encumbered by the Security Instrument are all legal parcels lawfully created in compliance with all subdivision laws and ordinances and, at Borrower's sole cost, Lender shall have received any title insurance endorsements to that effect reasonably requested by Lender; and (ii) the portion of the Property which shall remain encumbered by the Security Instrument has the benefit of all utilities, easements, public and/or private streets, covenants, conditions and restrictions as may be necessary, in Lender's reasonable opinion, for the anticipated development, improvement and use thereof in accordance with the Plans and Specifications, the Development Agreement and the Lease; and
- (b) not more than ten (10) days after such partial release of the Security Instrument, such roadway shall be dedicated and/or conveyed by Borrower to such Governmental Authority.
- Restrictions on Easements, Covenants and Restrictions. Borrower will not create or suffer to be created or to exist any easement, right of way, restriction, covenant, condition, license, cross easement, reciprocal parking agreement, or other right in favor of any Person or entity which affects or might affect title to the Property or the use and occupancy of the Property or any part thereof without (i) submitting to Lender the proposed instrument creating such easement, right of way, covenant, condition, license or other right, accompanied by a survey in recordable form showing the exact location thereof and such other information as Lender may reasonably request and (ii) obtaining the prior written approval of Lender. With respect to utility easements which are required in connection with the completion of construction and/or operation of the Improvements, Lender agrees not to unreasonably withhold, condition or delay its consent and agrees to provide, and cause the Trustee under the Security Instrument to provide, any subordination agreement

subordinating the lien of the Security Instrument to such utility easements, in a form reasonably acceptable to Lender, which may be required in connection with Borrower's grant of such utility easements to the applicable utility provider.

7.17 <u>Borrower Not Obligated to Enter into Swap Agreements.</u> It is understood and agreed that Borrower is not obligated to enter into any Swap Agreements in connection with the Loan.

[Signatures on the Following Page]

IN WITNESS WHEREOF, the parties have caused this Construction Loan Agreement to be executed and delivered by their respective authorized officers as of the date of this Agreement.

BORROWER:

RIVERBEND OLD STATESVILLE, LLC a North Carolina limited liability company

By: /s/Anthony J. Galici

Name: Anthony J. Galici

Title: Executive Vice President

LENDER:

JPMORGAN CHASE BANK, N.A.

By: /s/Ryan Dempsey

Name: Ryan Dempsey

Title: Authorized Officer

EXHIBIT A: LEGAL DESCRIPTION

That certain real property located in Mecklenburg County, North Carolina more particularly described as follows:

PARCEL ONE

Lying and being situate in Mecklenburg County, North Carolina, and being more particularly described as follows:

COMMENCING at an existing concrete monument with North Carolina Grid Coordinates N:589020.31 E:1455703.90; THENCE with a tie line North 19 degrees 54 minutes 03 minutes West, a distance of 35.54 feet to a calculated point and POINT OF BEGINNING marking the northern line of Carrier Corporation & Tax Department (now or formerly) as described in Deed Book 9970, Page 918 of the Mecklenburg County Registry;

THENCE with the northern line of the aforementioned Carrier Corporation & Tax Department the following five (5) courses and distances 1) North 88 degrees 09 minutes 35 seconds West, a distance of 91.49 feet to a point; 2) THENCE along a curve to the right having an arc length of 104.94 feet, and a radius of 1050.00 feet, being subtended by a chord bearing of North 85 degrees 17 minutes 48 seconds West, a distance of 104.90 feet to a point; 3) THENCE North 82 degrees 26 minutes 00 seconds West, a distance of 170.43 feet to a point; 4) THENCE along a curve to the left having an arc length of 45.20 feet, and a radius of 1350.00 feet, being subtended by a chord bearing of North 83 degrees 23 minutes 33 seconds West, a distance of 45.20 feet to a point; 5) THENCE North 84 degrees 21 minutes 06 seconds West, a distance of 266.69 feet to a point marking the eastern line of Oldcastle APG South Inc (now or formerly) as described in Deed Book 28263, Page 828; THENCE with the eastern line of the aforementioned Oldcastle APG South Inc and HS-Tampa FL LLC & Property Tax Department (now or formerly) as described in Deed Book 17235, Page 382 North 05 degrees 31 minutes 39 seconds East, a total distance of 413.71 feet to an existing concrete monument marking the southernmost corner of Norris Inc (now or formerly) as described in Deed Book 5721, Page 182, passing an existing #5 rebar marking the southern margin of an unopened 60' right of way, as shown in Map Book 22, Page 29 at a distance of 21.17 feet and passing an existing #6 rebar marking the southern line of the aforementioned HS-Tampa FL LLC & Property Tax Department at an additional distance of 60.75 feet; THENCE with the southern line of the aforementioned Norris Inc and the southern line of Riverbend Old Statesville, LLC (now or formerly) as described in Deed Book 33672, Page 209 the following two (2) courses and distances 1) North 62 degrees 32 minutes 53 seconds East, a distance of 438.80 feet to an existing concrete monument passing an existing #5 rebar at a distance of 110.73 feet; 2) THENCE South 19 degrees 54 minutes 03 seconds East, a distance of 722.66 feet to the POINT OF BEGINNING, having an area of 6.874 acres, more or less.

PARCEL TWO

Lying and being situate in Mecklenburg County, North Carolina, and being more particularly described as follows:

COMMENCING at an existing concrete monument with North Carolina Grid Coordinates N:589020.31 E:1455703.90; THENCE with a tie line North 19 degrees 54 minutes 03 minutes West, a distance of 35.54 feet to a calculated point and POINT OF BEGINNING marking the southeastern corner of Riverbend Old Statesville, LLC (now or formerly) as described in Deed Book 33678, Page 169 of the Mecklenburg County Registry;

THENCE with line of the aforementioned Riverbend Old Statesville, LLC the following two (2) courses and distances 1) North 19 degrees 54 minutes 03 seconds West, a distance of 722.66 feet to an existing concrete monument; 2) THENCE South 62 degrees 32 minutes 53 seconds West, a distance of 328.07 feet to an existing #5 rebar marking the southeastern corner of Norris Inc (now or formerly) as described in Deed Book 5721, Page 182; THENCE with the eastern line of the aforementioned Norris Inc North 01 degrees 52 minutes 21 seconds West, a distance of 949.72 feet to an existing #5 rebar marking the southern line of SWD Investments LLC C/O Silas W Davis Jr (now or formerly) as described in Deed Book 27931, Page 640; THENCE South 69 degrees 10 minutes 22 seconds East, a distance of 1074.31 feet to an existing #5 rebar marking the northwestern corner of Mecklenburg County C/O Real Estate/Finance Dept (now or formerly) as described in Deed Book 2940, Page 318; THENCE with the southern line of the aforementioned Mecklenburg County C/O Real Estate/Finance Dept

the following two (2) courses and distance 1) South 69 degrees 10 minutes 22 seconds East, a distance of 111.59 feet to a calculated point in a concrete lid at spring; THENCE South 87 degrees 13 minutes 42 seconds East, a distance of 397.91 feet to a calculated point marking the centerline of Old Statesville Road; THENCE with the centerline of Old Statesville Road the following ten (10) courses and distances 1) South 05 degrees 28 minutes 16 seconds East, a distance of 288.50 feet; 2) THENCE South 04 degrees 12 minutes 39 seconds East, a distance of 100.00 feet to a point; 3) THENCE South 01 degrees 19 minutes 48 seconds East, a distance of 88.17 feet to a point; 4) THENCE South 00 degrees 09 minutes 06 seconds West, a distance of 100.00 feet to a point 5) THENCE South 02 degrees 30 minutes 30 seconds West, a distance of 100.01 feet to a point; 6) THENCE South 04 degrees 11 minutes 28 seconds West, a distance of 100.00 feet to a point; 7) THENCE South 06 degrees 17 minutes 04 seconds West, a distance of 99.97 feet to a point; 8) THENCE South 07 degrees 19 minutes 33 seconds West, a distance of 82.76 feet; feet to a point 9) THENCE South 07 degrees 56 minutes 12 seconds West, a distance of 144.95 feet to a point; 10) THENCE South 07 degrees 36 minutes 46 seconds West, a distance of 44.94 feet to a point marking the northeastern corner of Carrier Corporation & Tax Department (now or formerly) as described in Deed Book 9970, Page 918; THENCE with the northern line of the aforementioned Carrier Corporation & Tax Department the following six (6) courses and distances 1) North 81 degrees 59 minutes 15 seconds West, a distance of 325.35 feet; 2) THENCE North 79 degrees 17 minutes 50 seconds West, a distance of 85.22 feet;3) THENCE North 81 degrees 59 minutes 15 seconds West, a distance of 229.02 feet;4) THENCE along a curve to the left having an arc length of 62.60 feet, and a radius of 952.47 feet, being subtended by a chord bearing of North 83 degrees 52 minutes 28 seconds West, a distance of 62.59 feet to a; 5) THENCE with a compound curve to the left having an arc length of 39.74 feet, and a radius of 950.00 feet, being subtended by a chord bearing of North 86 degrees 57 minutes 40 seconds West, a distance of 39.74 feet to a; 6) THENCE North 88 degrees 09 minutes 35 seconds West, a distance of 180.67 feet to the POINT OF BEGINNING, having an area of 36.572 acres, more or less.

EXHIBIT B: CLOSING CONDITIONS

- (a) Loan Documents and Certain Third Party Documents. Lender shall have received on or before the date of this Agreement, the following documents fully executed and in form and substance satisfactory to Lender: (i) Loan Documents; (ii) Borrower Financing Statement; (iii) BAQ; (iv) the Contractor Subordination Agreement; (v) the Architect Subordination Agreement; and (vi) Engineer Subordination Agreement.
- (b) <u>Additional Closing Deliveries</u>. Lender shall have received on or before the date of this Agreement, the following in form and substance satisfactory to Lender:
 - (i) An opinion or opinions from counsel for Borrower and Guarantor covering such items as Lender and its counsel may reasonably require;
 - (ii) Copies of organizational documents for Borrower, Guarantor and any other entity reasonably requested by Lender based on the organizational chart for Borrower, together with good standing certificates (or their equivalent in the jurisdiction of organization of the applicable entity), resolutions, incumbency certificates, and such other evidence of authority to enter into the Loan Documents as Lender may reasonably require;
 - (iii) Current UCC, tax and judgment searches dated no earlier than thirty (30) days prior to the date of this Agreement made in such places as Lender may reasonably specify, covering Borrower and showing no filings relating to, or which could relate to, the Property other than those made under this Agreement;
 - (iv) Evidence of the insurance required under Section 5.01;
 - (v) A commitment to issue the Title Policy, together with copies of all documentation evidencing exceptions therein;
 - (vi) An ALTA/NSPS survey of the Property certified in a manner acceptable to Lender (the "Survey");
 - (vii) Evidence satisfactory to Lender that the Loan-to-Value Ratio is no more than fifty percent (50%);
 - (viii) An Appraisal of the Property;
 - (ix) The most recent available financial statements of Borrower and Guarantor;
 - (x) Evidence of satisfaction of the Flood Insurance Requirements;
 - (xi) An environmental report with respect to the Property prepared by an environmental consultant acceptable to Lender;
 - (xii) Reserved.

- (xiii) Evidence indicating compliance by the Improvements with applicable zoning requirements (without requirement for a variance);
- (xiv) Evidence that all utilities and municipal services (other than electricity and gas) required for the construction and operation of the Improvements are available at the Property;
- (xv) Copies of the executed Amazon Lease, together with a tenant estoppel certificate, and a fully executed subordination, non-disturbance and attornment agreement in form and substance acceptable to Lender for the Amazon Lease;
- (xvi) A Certification of Non-Foreign Status;
- (xvii) A signed IRS Form W-8 or W-9 as applicable;
- (xviii) Reserved;
- (xix) A geotechnical report with respect to the Property prepared by an engineer acceptable to Lender, and copies of all other inspection and test reports with respect to the Property made by, or for, Borrower;
- (xx) The Beneficial Ownership Certification; and
- (xxi) Copies of the executed Development Agreement,

Such other information and documents as Lender may reasonably require.

- (c) No Default. No Default or Unmatured Default shall have occurred and be continuing.
- (d) <u>Construction Documents</u>. Lender shall have received on or before the date of the Initial Advance, the following in form and substance satisfactory to Lender in its commercially reasonable judgment:
 - (i) The Plans and Specifications;
 - (ii) A plan and cost review report from the Inspecting Professional;
 - (iii) The Construction Schedule;
 - (iv) The Budget;
 - (v) The Architect's Certificate;
 - (vi) Reserved;
 - (vii) Evidence that all permits, certificates, licenses and approvals required under all applicable
 Legal Requirements for the construction of the Improvements, to the extent required given the stage of construction;

- (viii) A copy of the executed Construction Contract together with a schedule of values;
- (ix) Reserved.
- (x) Copies of such financial statements of the Contractor and each subcontractor as Lender may reasonably require, including balance sheets and profit and loss statements;
- (xi) A copy of the standard form of subcontract to be used by the Contractor, which form shall not prohibit an assignment of the subcontract to Lender or require the subcontractor's consent thereto and shall be used for all subcontracts; and
- (xii) Additional documents and information relating to the design and construction of the Improvements as required by Lender and consistent with the requirements in this Exhibit B and customarily required by construction lenders for projects similar to the Improvements

EXHIBIT C: ADVANCES

- (a) Equity Investment. On or before the date of the Initial Advance, Lender shall have received satisfactory evidence that (i) Borrower has invested equity in the Project in an amount not less than Thirteen Million Six Hundred Twenty-Four Thousand One Hundred Twenty and No/100 Dollars (\$13,624,120.00) (the "Borrower's Equity Requirement"), and (ii) Borrower has contributed the Total HVCRE Equity including the 15% Equity Contribution Requirement.
- (b) Borrowing Request. Lender shall have received a Borrowing Request.
- (c) <u>Draw Package</u>. Lender shall have received the following in form and substance satisfactory to Lender (collectively, a "<u>Draw Package</u>") at least ten (10) Business Days prior to the date of the requested Advance:
 - (i) a fully completed and executed request for loan advance in the form of <u>Schedule 1</u> attached:
 - (ii) a Project Summary and Transfer Request in form and detail acceptable to Lender in its commercially reasonable judgment;
 - (iii) a list of Soft Costs to be paid from the requested Advance, and copies of invoices for each item of Soft Costs:
 - (iv) an AIA Form G702 and G703 or similar form from the Contractor (with Contractor's sworn statement and application for payment attached);
 - (v) copies of invoices and conditional and unconditional partial lien waivers or releases of lien (or, in the case of an Advance for payment of final construction costs and release of retainage to the Contactor, an unconditional final lien waiver or release of lien) for all lienable work done and materials delivered;
 - (vi) Borrower shall require Contractor to provide to the Inspecting Professional: current copies of all executed change orders for the requested Advance, PCO and CO logs, RFI logs, buyout logs, executed subcontracts, subcontractor applications for payment, material testing logs, meeting minutes, lien waiver tracking logs and current updated construction schedule;
 - (vii) a current schedule of Stored Materials and a schedule of materials with deposits acceptable to Lender in its commercially reasonable judgment;
 - (viii) a list of Tenant Allowance Payments to be paid from the requested Advance, copies of the requests from tenants for the payment of such Tenant Allowance Payments and evidence that all of the conditions for such payment under the applicable Approved Lease have been satisfied;

- (ix) a list of Tenant Improvement Costs (other than Tenant Allowance Payments) to be paid from the requested Advance, and draw request certifications in the forms described in clause (iv) above from the contractor and architect for the applicable tenant improvement;
- (x) to the extent not previously delivered to Lender, copies of all permits, certificates, licenses and approvals required under applicable Legal Requirements for the construction of the Improvements as of the date of the requested Advance and copies of all subcontracts
- (xi) confirmation that all Project Costs to be funded with the requested Advance constitute "Allowable Development Costs" as defined and described in the Development Agreement and that the Draw Package has been approved or deemed approved by Amazon Tenant pursuant to and in accordance with the Development Agreement; and
- (xii) additional documentation reasonably requested by Lender.
- (d) <u>Inspecting Professional's Report</u>. Lender shall have received a report from the Inspecting Professional approving the subject Draw Package (including all Soft Costs).
- (e) <u>Title Endorsements</u>. If required to do so to have the title policy cover the amount of each new advance, Lender shall have received a commitment from the title company approved by Lender to issue an endorsement to the Title Policy extending the coverage to include the date and the amount of the requested Advance, without exception for any matter not previously approved by Lender in writing.
- (f) Loan In Balance. The Loan shall be "in balance."
- (g) <u>No Default or Unmatured Default</u>. No Default or Unmatured Default shall have occurred and be continuing; provided, however, that the absence of an Unmatured Default which will be cured by the requested Advance shall not be a condition to such Advance.
- (h) Representations and Warranties. The representations and warranties made under this Agreement or under any of the other Loan Documents, or in any certificate or other document executed by Borrower or Guarantor and delivered to Lender pursuant to, or in connection with, this Agreement, shall be true and correct in all material respects as of the applicable Borrowing Date except to the extent any such representation or warranty is made as of a specified date, in which case such representation or warranty shall have been true and correct on and as of such specified date.
- (i) <u>Amazon Lease</u>. The Amazon Lease remains in full force and effect and Borrower shall not have received any notice from Amazon Tenant of an election to terminate the Amazon Lease, whether pursuant to Section 4.06 or 4.07 of the Development Agreement or otherwise.

(j) <u>Utilities</u>. Prior to the Initial Advance of the Loan, Borrower shall deliver to Lender evidence that all electricity and gas utility services required for the construction and operation of the Improvements are available at the Property.

Each Draw Package submitted by Borrower shall constitute a representation and warranty by Borrower that, except as otherwise specifically disclosed in such Draw Package and labeled as a "Disclosure" (a "Disclosure"), all of the conditions to the applicable Advance in this Agreement have been and remain satisfied as of the applicable Borrowing Date. If Lender elects to make an Advance notwithstanding matters which are the subject of a Disclosure, the waiver of such matters shall be effective for that Advance only, and such matters must be corrected before the next Advance.

SCHEDULE 1

FORM OF BORROWING REQUEST

REQUEST FOR LOAN ADVANCE

| JPMorgan Chase Bank, N.A. Commercial Real Estate Loan Administration | |
|--|--|
| Draw #: Project Name: Requested Disbursement Date (which must be a B | Business Day): |
| from time to time, the "Agreement"), between Riverb | 7, 2021 (as amended, supplemented or otherwise modified bend Old Statesville, LLC, a North Carolina limited liability .A. (" <u>Lender</u> "). (Undefined terms used herein shall have the |
| below. Borrower acknowledges that these amount | to Section 2.02 of the Agreement in the amounts set forth ts are subject to inspection, verification, available funds and greement. Borrower understands that no funds shall be unt has been quoted by Lender. |
| Source | Requested Amount |
| Construction Loan | \$ |
| Borrower Equity | \$ |
| [Other] | \$ |
| 2. Borrower agrees: | |
| | g of all vendors showing the name and the amount currently d for labor, materials, and/or services supplied and (ii) that of this request. |
| Timing of the release of funds will be pursua draw documentation may result in delays to the | ant to the terms of the Agreement. Incorrect or incomplete ne timing of the release of funds. |
| Borrower certifies that the statements made in t Package are true and the undersigned is fully auth | his request and in any documents submitted in the Draw horized to execute this request on its behalf. |
| Date: | RIVERBEND OLD STATESVILLE, LLC |
| | By: (Must be an authorized signer) |
| | Printed Name: |

EXHIBIT D: LITIGATION AGAINST BORROWER, THE PROPERTY, OR GUARANTOR

NONE

EXHIBIT E: FORM OF BORROWER COMPLIANCE CERTIFICATE

[BORROWER'S LETTERHEAD] [Address]

| , 20 |
|---|
| COMPLIANCE CERTIFICATE |
| Via Email REB.Compliance@jpmorgan.com |
| JPMorgan Chase Bank, N.A. Attention: Ryan Dempsey |
| Ladies and Gentlemen: |
| This compliance certificate (this " <u>Certificate</u> ") is being delivered by Riverbend Old Statesville, LLC, a North Carolina limited liability company (" <u>Borrower</u> ") pursuant to <u>Section 3.08(a)(iii)</u> of that certain Construction Loan Agreement, dated as of May 7, 2021, by and between Borrower and Lender (as the same may have been or may hereafter be amended, modified, supplemented, restated and/or replaced from time to time, the " <u>Loan Agreement</u> "). All capitalized terms used but not defined in this Certificate shall have the meanings given in the Loan Agreement. |
| This Certificate is being given for the fiscal [quarter] [year] ending on, 20 (the "Covenant Certification Date"). Borrower hereby certifies to Lender as follows: |
| 1. The undersigned is currently the of Borrower and is fully authorized to act in such capacity and execute and deliver this Certificate. |
| 2. No Default or Unmatured Default has occurred and is continuing as of the date of this Certificate, except as set forth below [if blank, there are no exceptions]: |
| |
| 3. All representations and warranties made by Borrower in the Loan Documents (and any certificate, document or financial or any other statement furnished pursuant to, or in connection therewith) remain true and correct in all material respects (except for changes in conditions disclosed to Lender that would not cause a Default under the Loan Documents) on, and as of, the date of this Certificate with the same force and effect as if made on, and as of, such date. |
| 4. No material adverse effect as to Borrower has occurred since the date of the last compliance certificate delivered to Lender pursuant to Section[3.08(a)(iii)]) of the Loan Agreement, and no event or condition that could reasonably be expected to have a material adverse effect in the financial condition of Borrower has occurred, except as set forth below [if blank, there are no exceptions]: |
| |
| 5. Attached hereto as Exhibit A are the [quarterly] [annual] financial statements of Borrower required by Section 3.08[(a)(i)][(a)(ii) of the Loan Agreement for the fiscal [quarter] [year] ending on, 20, which financial statements are true, accurate and complete as of the date thereof. |

6. Attached hereto as Exhibit B are the quarterly operating statements, leasing reports and rent rolls required by <u>Section 3.08(a)(i)</u> of the Loan Agreement, which operating statements, leasing reports and rent rolls, are true, accurate and complete as of the date hereof.

[Remainder of Page Intentionally Left Blank; Signature Page Follows]

| IN WITNESS WHEREOF, the un | ndersigned has executed this Certificate this day of |
|----------------------------|--|
| | BORROWER: |
| | RIVERBEND OLD STATESVILLE, LLC, a North Carolina limited liability company |
| | Ву: |
| | Name: Title: |

EXHIBIT A

FINANCIAL STATEMENTS,

EXHIBIT B

OPERATING STATEMENTS AND LEASING REPORTS

EXHIBIT F: FORM OF GUARANTOR COMPLIANCE CERTIFICATE

[GUARANTOR'S LETTERHEAD] [Address]

| COMPLIANCE CERTIFICATE Via Email REB.Compliance@jpmorgan.com JPMorgan Chase Bank, N.A. Attention: Ryan Dempsey Ladies and Gentlemen: This compliance certificate (this "Certificate") is being delivered by INDUS Realty Trust, Inc., a Maryland corporation ("Guarantor") pursuant to Section 3.08(b)(iii) of that certain Construction Loan Agreement, dated as of May 7, 2021 by and between Borrower and Lender (as the same may have been or may hereafter be amended, modified, supplemented, restated and/or replaced from time to time, the "Loan Agreement"). All capitalized terms used but not defined in this Certificate shall have the meanings given in the Loan Agreement. This Certificate is being given for the fiscal [quarter] [year] ending on, 20 (the "Covenant Certification Date"). Guarantor hereby certifies to Lender as follows: 1. The undersigned is currently the of Guarantor and is fully authorized to act in such capacity and execute and deliver this Certificate. 2. No Default or Unmatured Default, has occurred and is continuing as of the date of this Certificate, except as set forth below [if blank, there are no exceptions]: |
|--|
| JPMorgan Chase Bank, N.A. Attention: Ryan Dempsey Ladies and Gentlemen: This compliance certificate (this "Certificate") is being delivered by INDUS Realty Trust, Inc., a Maryland corporation ("Guarantor") pursuant to Section 3.08(b)(iii) of that certain Construction Loan Agreement, dated as of May 7, 2021 by and between Borrower and Lender (as the same may have been or may hereafter be amended, modified, supplemented, restated and/or replaced from time to time, the "Loan Agreement"). All capitalized terms used but not defined in this Certificate shall have the meanings given in the Loan Agreement. This Certificate is being given for the fiscal [quarter] [year] ending on |
| Attention: Ryan Dempsey Ladies and Gentlemen: This compliance certificate (this "Certificate") is being delivered by INDUS Realty Trust, Inc., a Maryland corporation ("Guarantor") pursuant to Section 3.08(b)(iii) of that certain Construction Loan Agreement, dated as of May 7, 2021 by and between Borrower and Lender (as the same may have been or may hereafter be amended, modified, supplemented, restated and/or replaced from time to time, the "Loan Agreement"). All capitalized terms used but not defined in this Certificate shall have the meanings given in the Loan Agreement. This Certificate is being given for the fiscal [quarter] [year] ending on |
| This compliance certificate (this "Certificate") is being delivered by INDUS Realty Trust, Inc., a Maryland corporation ("Guarantor") pursuant to Section 3.08(b)(iii) of that certain Construction Loan Agreement, dated as of May 7, 2021 by and between Borrower and Lender (as the same may have been or may hereafter be amended, modified, supplemented, restated and/or replaced from time to time, the "Loan Agreement"). All capitalized terms used but not defined in this Certificate shall have the meanings given in the Loan Agreement. This Certificate is being given for the fiscal [quarter] [year] ending on, 20 (the "Covenant Certification Date"). Guarantor hereby certifies to Lender as follows: 1. The undersigned is currently the of Guarantor and is fully authorized to act in such capacity and execute and deliver this Certificate. 2. No Default or Unmatured Default, has occurred and is continuing as of the date of this |
| Maryland corporation ("Guarantor") pursuant to Section 3.08(b)(iii) of that certain Construction Loan Agreement, dated as of May 7, 2021 by and between Borrower and Lender (as the same may have been or may hereafter be amended, modified, supplemented, restated and/or replaced from time to time, the "Loan Agreement"). All capitalized terms used but not defined in this Certificate shall have the meanings given in the Loan Agreement. This Certificate is being given for the fiscal [quarter] [year] ending on |
| "Covenant Certification Date"). Guarantor hereby certifies to Lender as follows: 1. The undersigned is currently the of Guarantor and is fully authorized to act in such capacity and execute and deliver this Certificate. 2. No Default or Unmatured Default, has occurred and is continuing as of the date of this |
| in such capacity and execute and deliver this Certificate.2. No Default or Unmatured Default, has occurred and is continuing as of the date of this |
| , |
| |
| |
| 3. All representations and warranties made by Guarantor in the Loan Documents (and any certificate, document or financial or any other statement furnished pursuant to or in connection therewith) remain true and correct in all material respects (except for changes in conditions disclosed to Lender that would not cause a Default under the Loan Documents) on and as of the date of this Certificate with the same force and effect as if made on and as of such date. |
| 4. No material adverse effect as to Guarantor has occurred since the date of the last compliance certificate delivered to Lender pursuant to Section 3.08(b)(iii) of the Loan Agreement, and no event or condition that could reasonably be expected to have a material adverse effect in the financial condition of Guarantor has occurred, except as set forth below [if blank, there are no exceptions]: |
| |

5. Attached hereto as Exhibit [A] are the [quarterly] [annual] financial statements of Guarantor required by Section [3.08(b)(i)][3.08(b)(ii)] of the Loan Agreement for the fiscal [quarter] [year] ending on ______, 20___, which financial statements are true, accurate and complete as of the date thereof.

[Remainder of Page Intentionally Left Blank; Signature Page Follows]

| ~£ | IN WITNESS WHEREOF, the undersigned has executed this Certificate this | day |
|----|--|-----|
| | <u>GUARANTOR</u> : | |
| | INDUS REALTY TRUST, INC., a Maryland corporation | |
| | By: Name: Title: | |

EXHIBIT A

FINANCIAL STATEMENTS

EXHIBIT G: BUDGET

| LAND | |
|--------------------------------|---------------------|
| Land | \$ 7,250,000.00 |
| Lease Costs | \$ 400,000.00 |
| | |
| SUB-TOTAL LAND | \$ 7,650,000.00 |
| HARD COST | |
| GC Contract | \$ 27,130,028.00 |
| Off-Site Improvements (Not GC) | \$ 1,779,451.00 |
| | 4.050.500.00 |
| Hard Cost Contingency | \$ 1,253,589.00 |
| SUB-TOTAL HARD COST | \$ 30,163,068.00 |
| SOFT COST_ | |
| Engineer | \$ 245,000.00 |
| Architect | \$ 475,255.00 |
| Muni Fess & Permits | \$ 274,485.00 |
| Developer OH/Fee | \$ 1,216,312.00 |
| QC Special Testing | \$ 200,000.00 |
| Tenant Developer Consultant | \$ 600,000.00 |
| Utilities | \$ 50,000.00 |
| Security Services/Construction | \$ 30,000.00 |
| Injsurance | \$ 50,000.00 |
| Developer Travel Expenses | \$ 20,000.00 |
| Loan Fee/Origination | \$ 336,000.00 |
| | \$ - |
| Soft Cost Contingency | \$ 500,000.00 |
| SUB-TOTAL SOFT COST | \$ 3,997,052.00 |
| INTEREST RESERVE | |
| Interest Reserve | \$ 214,000.00 |
| SUB-TOTAL INTEREST | \$ 214,000.00 |
| L/C SUBLIMITS | |
| L/C Sublimit | \$ - |
| SUB TOTAL L/C SUBLIMITS | \$ - |
| PROJECT BUDGET | \$ 42,024,120.00 |
| LESS UPFRONT EQUITY | \$ 13,624,120.00 |
| LESS DEFERRED EQUITY | \$ - |
| LOAN TOTAL | \$ 28,400,000.00 |

NAI-1516568741v8

NEWS FROM: INDUS REALTY TRUST, INC.

CONTACT: Anthony Galici Chief Financial Officer (860) 286-1307 agalici@indusrt.com

Ashley Pizzo
Director, IR & Capital Markets
(212) 218-7914
apizzo@indusrt.com

INDUS ANNOUNCES FISCAL 2021 FIRST QUARTER RESULTS AND QUARTERLY DIVIDEND

NEW YORK, NEW YORK (May 10, 2021) INDUS Realty Trust, Inc. (Nasdaq: INDT) ("INDUS" or the "Company") today reported financial results for the three months ended March 31, 2021 (the "2021 first quarter") and the month of December 2020. In connection with its anticipated election to become a real estate investment trust ("REIT") for the year ending December 31, 2021, the Company changed its fiscal year from November 30 to December 31 effective with the 2021 fiscal year that began on January 1, 2021. As a result of this change, INDUS had a one-month transition period of December 2020 (the "Transition Period").

2021 First Quarter & Recent Highlights

- Net loss of \$0.8 million for the 2021 first quarter compared to a net loss of \$0.3 million for the three months ended March 31, 2020 (the "2020 first quarter")
- Net Operating Income ("NOI")¹ of \$7.0 million for the 2021 first quarter as compared to \$6.3 million for the 2020 first quarter
- Cash NOI² for industrial/logistics properties of \$5.9 million for the 2021 first quarter, as compared to \$5.2 million for the 2020 first quarter
- Completed a public offering of 1,927,049 shares of common stock for net proceeds of \$108.7 million, after expenses
- Industrial/logistics portfolio was 99.2% leased as of March 31, 2021; subsequently increased to 99.4% leased in April 2021
- Signed approximately 202,000 square feet of first generation leases on in-service industrial/logistics portfolio
- Signed approximately 297,000 square feet of first generation leases for industrial/logistics buildings in its development pipeline
- Subsequent to quarter end, entered into a construction loan agreement to provide a portion of the funds for the
 development costs of the build-to-suit in Charlotte, North Carolina
- Subsequent to quarter end, entered into an agreement to acquire a fully leased approximately 127,500 square foot industrial/logistics building in the Lehigh Valley of Pennsylvania
- Subsequent to quarter end, closed on the acquisition of approximately 14 acres of undeveloped land in Orlando, Florida for \$5.25 million
- Announced total potential proceeds of approximately \$44.0 million from sales of non-core properties currently under agreement (\$26.0 million) and the sale of an industrial/logistics building in Connecticut under agreement (\$18.0 million), if all sales were to close
- Declared a quarterly dividend on its common stock of \$0.15 per share payable on June 30, 2021 to holders of record as of the close of business on June 16, 2021

Quarterly Dividend

¹ NOI is not a financial measure in conformity with generally accepted accounting principles in the United States of America ("U.S. GAAP"). For additional information, see "Note Regarding Non-GAAP Financial Measures."

² Cash NOI is not a financial measure in conformity with U.S. GAAP. For additional information, see "Note Regarding Non-GAAP Financial Measures."

On May 7, 2021, INDUS's Board of Directors declared a dividend on its common stock of \$0.15 per share payable on June 30, 2021 to holders of record as of the close of business on June 16, 2021. The cash dividend is intended to be paid quarterly and represents an annualized dividend rate of \$0.60 per share. Prior to 2021, INDUS typically paid an annual cash dividend, but in connection with the Company's election to be taxed as a REIT, INDUS has updated its dividend policy to commence payment of a quarterly cash dividend. The Company will revisit the amount of its dividend from time to time in accordance with the growth in its earnings.

2021 First Quarter Results of Operations

INDUS reported total rental revenue of approximately \$10.1 million for the 2021 first quarter, as compared to approximately \$8.9 million for the 2020 first quarter. The approximately \$1.2 million increase in rental revenue was principally due to new leases of first generation space that commenced subsequent to the end of the 2020 first quarter and, to a lesser extent, leases of previously vacant second generation space, an increase in expense reimbursements and renewals on leases that drove rent increases subsequent to the end of the 2020 first quarter.

Net Operating Income ("NOI"), which is defined as rental revenue less operating expenses of rental properties and real estate taxes, increased to approximately \$7.0 million in the 2021 first quarter, from approximately \$6.3 million in the 2020 first quarter. The increase in NOI reflected the increase in rental revenue, as noted above, partially offset by an increase in operating expenses of rental properties principally due to higher snow removal costs.

Cash NOI for the 2021 first quarter was approximately \$6.6 million, as compared to approximately \$5.8 million for the comparable prior year period. The \$0.8 million increase in Cash NOI principally reflects an increase in rental revenue partially offset by the increase in operating expenses, as noted above.

NOI and Cash NOI for INDUS's industrial/logistics properties and total portfolio were as follows:

| (\$ in 000s) | | For the Three Months Ended | | | _ | Fo | r the | Month Ended | | |
|---------------------------------|----|----------------------------|----|---------------|----------|----------|---------------|-------------|---------------|----------|
| | _ | Mar. 31, 2021 | ľ | Mar. 31, 2020 | Increase | <u>.</u> | Dec. 31, 2020 | | Dec. 31, 2019 | Increase |
| Industrial/logistics portfolio: | | | | | | | | | | |
| NOI | \$ | 6,329 | \$ | 5,531 | 14.4% | \$ | 2,114 | \$ | 1,777 | 19.0% |
| Cash NOI | \$ | 5,933 | \$ | 5,222 | 13.6% | \$ | 1,882 | \$ | 1,662 | 13.2% |
| | | | | | | | | | | |
| Total portfolio: | | | | | | | | | | |
| NOI | \$ | 7,007 | \$ | 6,309 | 11.1% | \$ | 2,366 | \$ | 1,944 | 21.7% |
| Cash NOI | \$ | 6,570 | \$ | 5,777 | 13.7% | \$ | 2,116 | \$ | 1,787 | 18.4% |

General and administrative expenses increased to approximately \$3.0 million in the 2021 first quarter from approximately \$2.1 million in the 2020 first quarter, principally reflecting a \$0.8 million increase in expense related to INDUS's non-qualified deferred compensation plan.

INDUS incurred a net loss of approximately \$0.8 million in the 2021 first quarter, as compared to a net loss of approximately \$0.3 million for the 2020 first quarter. In addition to the items described above, contributing to the higher net loss was a decrease of approximately \$0.6 million in gain on sales of real estate assets, partially offset by the inclusion in the 2021 first quarter of an approximately \$0.3 million gain from the change in the fair value of financial instruments that were issued on August 24, 2020 in connection with the private placement of the Company's common stock completed at that time.

Leasing Activity

During the 2021 first quarter, INDUS executed four first generation leases totaling approximately 202,000 square feet at 160 and 180 International Drive in the Charlotte market and 170 Sunport Lane in the Orlando market. These new leases had a weighted average lease term of 6.5 years and a weighted average lease cost per square foot per year of \$1.19.³ Additionally, including a lease signed subsequent to the 2021 first quarter, the Company's most recent value-add acquisition, 170 Sunport Lane, is over 60% leased with only approximately 27,000 square feet of vacancy remaining.

³ Weighted average lease cost per square foot per year reflects total lease costs (tenant improvements, leasing commissions and legal costs) per square foot per year of the lease term.

Additionally, during the 2021 first quarter, INDUS executed two first generation leases totaling approximately 297,000 square feet for projects currently in its development pipeline (see below section on "Acquisition & Development Pipeline"). One lease is for a fifteen-year lease agreement and development agreement with Amazon for a build-to-suit development (the "Charlotte Build-to-Suit") on the Company's 44 acre land parcel located on Old Statesville Road in Charlotte (the "Charlotte Land"). The other lease is for a seven-year agreement with a leading global shipping and logistics company for a portion of a to-be-constructed approximately 234,000 square foot industrial/logistics building on the Company's 16 acre land parcel at 110 Tradeport Drive (the "110 Tradeport Development") in New England Tradeport, the Company's industrial park in Windsor and East Granby, Connecticut.

During the Transition Period, INDUS executed a 12-month renewal for an approximately 228,000 square foot full building lease in the Lehigh Valley with a third-party logistics company that was originally scheduled to expire on September 30, 2021.

As of March 31, 2021, INDUS's thirty industrial/logistics buildings aggregated approximately 4,206,000 square feet and represented 91.5% of INDUS's total real estate portfolio. As a result of the activity described above, INDUS's in-service industrial/logistics portfolio's percentage leased was as follows:

| | Mar 31, 2021 | Dec 31, 2020 | Aug 31, 2020 | May 31, 2020 |
|--|-----------------|-----------------|-----------------|-----------------|
| Percentage Leased | 99.2% | 94.5% | 94.3% | 94.3% |
| Percentage Leased – Stabilized Properties ⁴ | 99.2% | 95.7% | 99.7% | 99.7% |

No new office/flex leasing was completed during the Transition Period or the 2021 first quarter. INDUS's eleven office/flex buildings, which aggregate approximately 393,000 square feet and comprise 8.5% of INDUS's total real estate portfolio, were 71.3% leased as of March 31, 2021, unchanged from December 31, 2020.

Acquisition & Development Pipeline

Subsequent to March 31, 2021, INDUS entered into an agreement to purchase an approximately 127,500 square foot industrial/logistics building on approximately 13.7 acres of land in the Lehigh Valley for a purchase price of \$11.7 million (the "Lehigh Valley Acquisition"). The Lehigh Valley Acquisition is fully leased through December 2022 to a subsidiary of a publicly traded multinational chemical company and has a 4.5% in-place cash capitalization rate (first full year Cash NOI/purchase price). The Lehigh Valley Acquisition has excess, unutilized land that INDUS believes could receive approvals to be used for additional parking, for outdoor storage or to expand the existing building. The Company expects the Lehigh Valley Acquisition to close by May 31, 2021.

⁴ Stabilized Properties reflect buildings that have reached 90% leased or have been in service for at least one year since development completion or acquisition date, whichever is earlier. 170 Sunport Lane, which was 53.4% leased as of March 31, 2021, was acquired in March 2020 and is now included in the Stabilized Properties pool for the 2021 first quarter.

The following is a summary of INDUS's development pipeline for its industrial/logistics portfolio as of May 7, 2021, which includes the closing on April 13, 2021 of the purchase of an approximately 14 acre parcel of undeveloped land in Orlando for \$5.25 million (the "Jetport Land"), a portion of which was funded using proceeds of Section 1031 like-kind exchanges from previous non-core asset sales:

| | | Building | | Expected |
|-----------------------------------|-------------------|-----------|----------------|----------|
| Name | Market | Size (SF) | Type | Delivery |
| Owned Land | | | | |
| Charlotte Build-to-Suit | Charlotte, NC | 141,000 | Build-to-Suit | Q3 2021 |
| Chapmans Road | Lehigh Valley, PA | 103,000 | Speculative | Q4 2021 |
| 110 Tradeport Development | Hartford, CT | 234,000 | 67% Pre-leased | Q2 2022 |
| Jetport Land | Orlando, FL | 195,000 | Speculative | Q3 2022 |
| Land Under Purchase & Sale Agreem | ent | | | |
| First & Second Allentown Purchase | | | | |
| Agreements | Lehigh Valley, PA | 206,000 | Speculative | Q4 2022 |
| Total | | 879,000 | | • |

INDUS expects that the total development and stabilization costs of developments in its pipeline will total approximately \$113.5 million, of which approximately \$20.6 million has been expended through April 13, 2021. The Company has underwritten a weighted average stabilized Cash NOI yield between 6.1% - 6.6% on its development pipeline. Included in this total is the 110 Tradeport Development which will benefit from the low cost basis of the already entitled 16 acre land parcel owned by INDUS. The Company estimates the 110 Tradeport Development will generate an underwritten stabilized Cash NOI yield between 7.7% - 8.1%.⁵ Actual initial full year stabilized Cash NOI yields may vary from INDUS's underwritten stabilized Cash NOI yield ranges based on the actual total cost to complete a project or acquire a property and its actual initial full year stabilized Cash NOI.

Dispositions

Subsequent to the end of the 2021 first quarter, INDUS entered into agreements to sell a number of non-core properties, in addition to land placed under agreement during the quarter. These include agreements to sell approximately 217,000 square feet from INDUS's office/flex portfolio (which represents over 55% of its remaining office/flex square footage as of March 31, 2021), in addition to the agreement to sell one of the Company's specialized industrial buildings totaling 165,000 square feet in Windsor, CT to the user of the building (see below). The properties under agreement for sale have a combined mortgage balance of approximately \$9.1 million as of March 31, 2021, which INDUS intends to repay with proceeds from the sales of these properties.

On April 20, 2021, INDUS entered into the third amendment to the lease with the tenant in its approximately 7,200 square foot restaurant building in the Hartford market (included in the office/flex portfolio). Under the terms of this amendment, the tenant exercised its option to purchase the restaurant building for approximately \$0.6 million with the closing to take place on or before May 31, 2021.

On April 28, 2021, INDUS entered into an agreement (the "Windsor Office Sale Agreement") to sell: (a) 5 and 7 Waterside Crossing, two adjacent multi-story office buildings aggregating approximately 161,000 square feet; (b) 21 Griffin Road North, an approximately 48,000 square foot office/flex building; and (c) 25 Griffin Road North, an approximately 8 acre parcel of undeveloped land, for a total purchase price of \$6.6 million.

On April 29, 2021, INDUS entered into an agreement (the "Blue Hills Sale Agreement") with the full-building tenant in 1985 Blue Hills Avenue ("1985 Blue Hills"), an approximately 165,000 square foot industrial/logistics building in the

amortization), from its estimate of its initial full year stabilized rental revenue.

⁵ As a part of INDUS's standard development and acquisition underwriting process, INDUS analyzes the targeted initial full year stabilized Cash NOI yield for each development project and acquisition target and establishes a range of initial full year stabilized Cash NOI yields, which it refers to as "underwritten stabilized Cash NOI yields." Underwritten stabilized Cash NOI yields are calculated as a development project's or acquisition's initial full year stabilized Cash NOI as a percentage of its estimated total investment, including costs to stabilize the buildings to 95% occupancy (other than in connection with build-to-suit development projects and single tenant properties). INDUS calculates initial full year stabilized Cash NOI for a development project or acquisition by subtracting its estimate of the development project's or acquisition's initial full year stabilized operating expenses, real estate taxes and non-cash rental revenue, including straight-line rents (before interest, income taxes, if any, and depreciation and

Hartford market, to sell 1985 Blue Hills and two adjacent parcels of undeveloped land aggregating approximately 39 acres to the tenant for a purchase price of \$18.0 million. Under the terms of the Blue Hills Sale Agreement, closing on the sale is to take place upon 30 days written notice from INDUS to the buyer, but in no event shall it be later than December 15, 2021

In summary, as of May 7, 2021, INDUS has entered into agreements to sell the following non-core buildings and undeveloped land parcels:

| | | | | Expected | Sale Price |
|---|------------------------|-------------------|----------------------|----------|------------------|
| Name | Type | Location | Property Size | Closing | (\$ in millions) |
| 1936 Blue Hills Ave | Office/Flex | Windsor, CT | 7,199 SF | Q2 2021 | \$0.6 |
| 5 & 7 Waterside Crossing, 21 & 25 | | | 209,390 SF | | |
| Griffin Road North | Office/Flex | Windsor, CT | and 8 acres | Q3 2021 | \$6.6 |
| | | | 165,000 SF | | |
| 1975, 1985 & 1995 Blue Hills Ave | Industrial & Land | Windsor, CT | and 39 acres | Q4 2021 | \$18.0 |
| Subtotal Gross Proceeds of Proper | ty Dispositions Under | Agreement, if Con | summated | | \$25.2 |
| | | | | | |
| Floydville Road Lot #13 | Land | East Granby, CT | 8 acres | Q3 2021 | \$0.1 |
| Florida Nursery Farm | Land | Quincy, FL | 1,066 acres | Q3 2021 | \$1.1 |
| Southwick, MA Land | Land | Southwick, MA | 91 acres | Q3 2021 | \$5.2 |
| Stratton Farms Residential Parcels ⁶ | Land | Suffield, CT | 6 acres (7 lots) | Q3 2021 | \$0.4 |
| 60 Griffin Road South Land | Land | Bloomfield, CT | 34 acres | Q3 2021 | \$0.6 |
| Meadowood Residential Parcels | Land | Simsbury, CT | 277 acres | Q4 2021 | \$5.4 |
| | | East Granby / | | | |
| East Granby / Windsor Parcels | Land | Windsor, CT | 280 acres | 2022 | \$6.0 |
| Total Gross Proceeds of Land & P | roperty Dispositions U | Inder | | | \$44.0 |
| Agreement, if Consummated | | | | | |

The completion of the sales contemplated under these agreements is subject to satisfactory completion of due diligence by the buyers, among other contingencies. There can be no guarantee that the transactions contemplated will be completed under their current terms, or at all.

Liquidity & Capital Resources

On February 2, 2021, INDUS filed a universal shelf registration statement on Form S-3 (the "Universal Shelf") with the Securities and Exchange Commission. Under the Universal Shelf, INDUS may offer and sell up to \$500 million of a variety of securities during the three year period that commenced upon the effective date of the Universal Shelf. On March 5, 2021, under its Universal Shelf, INDUS completed an underwritten public offering of 1,750,000 shares of its common stock at a price to the underwriters of \$56.85 per share. On March 15, 2021, the underwriters exercised their option to purchase an additional 177,049 shares of common stock from INDUS at the same offering price. INDUS received total net proceeds of approximately \$108.7 million, after expenses, from the sale of its common stock and intends to use the proceeds to finance its development pipeline and acquisitions and for other corporate purposes.

On May 7, 2021, a subsidiary of INDUS entered into a construction loan agreement (the "2021 JPM Construction Loan") with JPMorgan Chase Bank N.A. to provide a portion of the funds for the development costs of the Charlotte Build-to-Suit. Total borrowings under the JPM Construction Loan will be the lesser of \$28.4 million or 67.5% of the total cost (as defined) of the Charlotte Build-to-Suit. The term of the 2021 JPM Construction Loan is two years, with a one-year extension at the Company's option. Interest under the 2021 JPM Construction Loan, to be adjusted monthly, is one-month LIBOR plus 1.65%, reduced to one-month LIBOR plus 1.40% upon completion of the Charlotte Build-to-Suit and commencement of rental payments by Amazon.

⁶ The sale of the 16 Stratton Farms residential parcels for a total of approximately \$0.9 million is to be completed in two parts. The sale of the first 9 lots closed in February 2021 and accounted for approximately \$0.5 million of the gross sales price. The sale of the remaining 7 lots is expected to close in the 2021 third quarter and represents approximately \$0.4 million of the total gross sales price.

As of March 31, 2021, the Company maintained approximately \$182.0 million of liquidity which reflects approximately \$132.0 million of cash and cash equivalents (primarily from the 2021 first quarter public offering) as well as \$50.0 million of capacity under its revolving credit facilities. As of March 31, 2021, there were no borrowings outstanding under the Company's revolving credit facilities.

First Quarter Earnings Conference Call, Earnings Supplement and Investor Presentation

INDUS is hosting a live earnings conference call that will take place tomorrow, May 11, 2021 at 11:00 A.M. Eastern Time, to discuss its 2021 first quarter and Transition Period operating results. Supplemental materials containing additional financial and operating information will be available on INDUS's website at the start of the call. All investors and other interested parties are invited to either dial in to the call (to participate in live Q&A) or log in to a listen-only webcast which, together with the supplemental information, can be accessed via the Investors section of INDUS's website at www.indusrt.com/investors or by calling the following numbers:

PARTICIPANT DIAL IN (TOLL FREE): 1-866-777-2509 PARTICIPANT INTERNATIONAL DIAL IN: 1-412-317-5413

An archived recording of the webcast will be available for three months under the Investors section of INDUS's website at www.indusrt.com.

About INDUS

INDUS (formerly known as Griffin Industrial Realty, Inc.) is a real estate business principally engaged in developing, acquiring, managing and leasing industrial/logistics properties. INDUS owns 41 buildings totaling approximately 4.6 million square feet (including 30 industrial/logistics buildings aggregating approximately 4.2 million square feet) in Connecticut, Pennsylvania, North Carolina and Florida in addition to over 3,400 acres of undeveloped land.

Forward-Looking Statements:

This Press Release includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended. These forward-looking statements include INDUS's beliefs and expectations regarding future events or conditions including, without limitation, statements regarding INDUS's intention to elect to be taxed as a REIT, the completion of acquisitions and dispositions under agreements, construction and development plans and timelines, the estimated underwritten stabilized Cash NOI of the 110 Tradeport Development and Cash NOI yield estimates, expected total development and stabilization costs of developments in INDUS's pipeline, anticipated leasing activity, expectations regarding excess, unutilized land at the Lehigh Valley Acquisition, and expected capital availability and liquidity. Although INDUS believes that its plans, intentions and expectations reflected in such forward-looking statements are reasonable, it can give no assurance that such plans, intentions or expectations will be achieved. The projected information disclosed herein is based on assumptions and estimates that, while considered reasonable by INDUS as of the date hereof, are inherently subject to significant business, economic, competitive and regulatory uncertainties and contingencies, many of which are beyond the control of INDUS and which could cause actual results and events to differ materially from those expressed or implied in the forward-looking statements. Other important factors that could affect the outcome of the events set forth in these statements are described in INDUS's Securities and Exchange Commission filings, including the "Business," "Risk Factors" and "Forward-Looking Statements" sections in INDUS's Annual Report on Form 10-K for the fiscal year ended November 30, 2020 filed with the SEC on February 18, 2021. INDUS disclaims any obligation to update any forwardlooking statements as a result of developments occurring after the date of this press release except as required by law.

Note Regarding Non-GAAP Financial Measures:

The Company uses NOI, Cash NOI, NOI of Industrial/Logistics Properties and Cash NOI of Industrial/Logistics Properties, as supplemental non-GAAP performance measures. Management believes that the use of these measures combined with net income (loss) (which remains the Company's primary measure of performance), improves the understanding of the Company's operating results among the investing public and makes comparisons of operating results to other REITs more meaningful.

NOI is a non-GAAP measure that includes the rental revenue and operating expense directly attributable to the Company's real estate properties. NOI of Industrial/Logistics Properties is NOI excluding NOI for the Company's non-

industrial/logistics properties. The Company uses NOI and NOI of Industrial/Logistics Properties as supplemental performance measures because, in excluding real estate depreciation and amortization expense, general and administrative expenses, interest expense, gains (or losses) on the sale of real estate and other non-operating items, they provide a performance measure that, when compared year over year, captures trends in occupancy rates, rental rates and operating costs. The Company also believes that NOI and NOI of Industrial/Logistics Properties will be useful to investors as a basis to compare its operating performance with that of other REITs. However, because NOI and NOI of Industrial/Logistics Properties excludes depreciation and amortization expense and captures neither the changes in the value of the Company's properties that result from use or market conditions, nor the level of capital expenditures and leasing commissions necessary to maintain the operating performance of its properties (all of which have real economic effect and could materially impact the Company's results from operations), the utility of NOI and NOI of Industrial/Logistics Properties as measures of the Company's performance is limited. Other equity REITs may not calculate NOI or NOI of Industrial/Logistics Properties in a similar manner and, accordingly, the Company's NOI and NOI of Industrial/Logistics Properties may not be comparable to such other REITs' NOI. Accordingly, NOI and NOI of Industrial/Logistics Properties should be considered only as a supplement to net income (loss) as a measure of the Company's performance. NOI and NOI of Industrial/Logistics Properties should not be used as a measure of the Company's liquidity, nor is it indicative of funds available to fund the Company's cash needs. NOI and NOI of Industrial/Logistics Properties should not be used as a substitute for cash flow from operating activities in accordance with U.S. GAAP.

Cash NOI is a non-GAAP measure that the Company calculates by adding or subtracting non-cash rental revenue, including straight-line rental revenue, from NOI. Cash NOI of Industrial/Logistics Properties is Cash NOI excluding NOI for the Company's non-industrial/logistics properties. The Company uses Cash NOI and Cash NOI of Industrial/Logistics Properties, as supplemental performance measures. Cash NOI and Cash NOI of Industrial/Logistics Properties should not be used as measures of the Company's liquidity, nor are they indicative of funds available to fund the Company's cash needs. Cash NOI and Cash NOI of Industrial/Logistics Properties should not be used as a substitute for cash flow from operating activities computed in accordance with U.S. GAAP.

INDUS REALTY TRUST, INC. Consolidated Statements of Operations (dollars in thousands, except per share data) (unaudited)

| | For the Three Months Ended | | | For the Month Ended | | | | |
|---|----------------------------|---------|----|---------------------|----|-------------|-----|------------|
| | Mar. 31, 2021 | | M | ar. 31, 2020 | De | c. 31, 2020 | Dec | . 31, 2019 |
| Rental revenue | \$ | 10,087 | \$ | 8,862 | \$ | 3,345 | \$ | 3,083 |
| | | | | | | | | |
| Expenses: | | | | | | | | |
| Operating expenses of rental properties | | 1,633 | | 1,172 | | 497 | | 681 |
| Real estate taxes | | 1,447 | | 1,381 | | 482 | | 458 |
| Depreciation and amortization expense | | 3,343 | | 3,306 | | 1,122 | | 1,059 |
| General and administrative expenses | | 2,970 | | 2,143 | | 1,172 | | 629 |
| Total operating expenses | · | 9,393 | | 8,002 | | 3,273 | | 2,827 |
| | | | | | | | | |
| Other income (expense): | | | | | | | | |
| Interest expense | | (1,749) | | (1,840) | | (602) | | (589) |
| Change in fair value of financial instruments | | 260 | | _ | | 2,785 | | _ |
| Gain on sales of real estate assets | | 20 | | 584 | | _ | | _ |
| Other expense | | _ | | _ | | (281) | | _ |
| Investment income | | 7 | | 25 | | 7 | | 2 |
| | | (1,462) | | (1,231) | | 1,909 | | (587) |
| | | | | | | | | |
| (Loss) income before income tax benefit | | (768) | | (371) | | 1,981 | | (331) |
| Income tax benefit | | _ | | 85 | | _ | | 76 |
| Net (loss) income | \$ | (768) | \$ | (286) | \$ | 1,981 | \$ | (255) |
| | | | | | | | | |
| | | | | | | | | |
| Basic net (loss) income per common share | \$ | (0.12) | \$ | (0.06) | \$ | 0.35 | \$ | (0.05) |
| | | | | | | | | |
| Diluted net (loss) income per common share | \$ | (0.12) | \$ | (0.06) | \$ | 0.34 | \$ | (0.05) |

INDUS REALTY TRUST, INC. Non-GAAP Reconciliations – NOI and Cash NOI (dollars in thousands)

| | 2021 First Quarter | 2020 First Quarter | December 2020 | December 2019 |
|---|--------------------------|--------------------------|------------------|------------------|
| Net (loss) income | (\$ 768) | (\$ 286) | \$ 1,981 | (\$ 255) |
| Income tax benefit | - | (85) | - | (76) |
| Pretax (loss) income | (768) | (371) | 1,981 | (331) |
| Exclude: | | | | |
| Depreciation and amortization expense | 3,343 | 3,306 | 1,122 | 1,059 |
| General and administrative expenses | 2,970 | 2,143 | 1,172 | 629 |
| Interest expense | 1,749 | 1,840 | 602 | 589 |
| Change in fair value of financial instruments | (260) | - | (2,785) | - |
| Gain on sales of real estate assets | (20) | (584) | - | - |
| Other expense | - | - | 281 | - |
| Investment income | (7) | (25) | (7) | (2) |
| NOI | 7,007 | 6,309 | 2,366 | 1,944 |
| Noncash rental revenue including straight-line rents | (437) | (532) | (250) | (157) |
| Cash NOI | \$ 6,570 | \$ 5,777 | \$ 2,116 | \$ 1,787 |
| | | | | |
| NOI | \$ 7,007 | \$ 6,309 | \$ 2,366 | \$ 1,944 |
| Exclude: | | | | |
| Rental revenue from non-industrial/logistics properties | (1,443) | (1,535) | (477) | (509) |
| Operating expenses of non-industrial/logistics properties | 567 | 539 | 159 | 269 |
| Real estate taxes of non-industrial/logistics properties | 198 | 218 | 66 | 73 |
| NOI of industrial/logistics properties | 6,329 | 5,531 | 2,114 | 1,777 |
| Noncash rental revenue including straight-line rents of | , | • | , | ŕ |
| industrial/logistics properties | (396) | (309) | (232) | (115) |
| Cash NOI of Industrial/Logistics Properties | \$ 5,933 | \$ 5,222 | \$ 1,882 | \$ 1,662 |
| _ | | | | |